IDE Discussion Papers are preliminary materials circulated to stimulate discussions and critical comments

IDE DISCUSSION PAPER No. 324

Heterogeneous Multinational Firms and Productivity Gains from Falling FDI Barriers

Shawn ARITA and Kiyoyasu TANAKA*

February 2012

Abstract

During the past decade of declining FDI barriers, small domestic firms disproportionately contracted while large multinational firms experienced a substantial growth in Japan's manufacturing sector. This paper quantitatively assesses the impact of FDI globalization on intra-industry reallocations and aggregate productivity. We calibrate the firm-heterogeneity model of Eaton, Kortum, and Kramarz (2011) to micro-level data on Japanese multinational firms. Estimating the structural parameters of the model, we demonstrate that the model can strongly replicate the entry and sales patterns of Japanese multinationals. Counterfactual simulations show that declining FDI barriers lead to a disproportionate expansion of foreign production by more efficient firms relative to less efficient firms. A hypothetical 20% reduction in FDI barriers is found to generate a 30.7% improvement in aggregate productivity through market-share reallocation.

Keywords: Multinational firms, FDI, Firm heterogeneity, Investment Liberalization **JEL classification:** F10, F23, L25, R12, R30

^{*} *Corresponding author*: Shawn Arita; Department of Economics, University of Hawaii at Manoa; e-mail: aritas@hawaii.edu

The Institute of Developing Economies (IDE) is a semigovernmental, nonpartisan, nonprofit research institute, founded in 1958. The Institute merged with the Japan External Trade Organization (JETRO) on July 1, 1998. The Institute conducts basic and comprehensive studies on economic and related affairs in all developing countries and regions, including Asia, the Middle East, Africa, Latin America, Oceania, and Eastern Europe.

The views expressed in this publication are those of the author(s). Publication does not imply endorsement by the Institute of Developing Economies of any of the views expressed within.

INSTITUTE OF DEVELOPING ECONOMIES (IDE), JETRO 3-2-2, Wakaba, Mihama-ku, Chiba-shi Chiba 261-8545, JAPAN

©2012 by Institute of Developing Economies, JETRO No part of this publication may be reproduced without the prior permission of the IDE-JETRO.

Heterogeneous Multinational Firms and Productivity Gains from Falling FDI Barriers*

Shawn $ARITA^{\zeta}$

University of Hawaii at Manoa

Kiyoyasu TANAKA§

Institute of Developing Economies

February 2012

Abstract:

During the last decade of declining FDI barriers, small domestic firms disproportionately contracted while large multinational firms experienced a substantial growth in Japan's manufacturing sector. This paper quantitatively assesses the impact of FDI globalization on intra-industry reallocations and its implications for aggregate productivity. We calibrate the firm-heterogeneity model of Eaton, Kortum, and Kramarz (2011) to micro-level data on Japanese multinational firms facing fixed and variable costs of foreign production. Estimating the structural parameters of the model, we demonstrate that the model can strongly replicate the entry and sales patterns of Japanese multinationals. Counterfactual simulations show that declining FDI barriers lead to a disproportionate expansion of foreign production by more efficient firms relative to less efficient firms. A hypothetical 20% reduction in FDI barriers is found to generate a 30.7% improvement in aggregate productivity through market-share reallocation.

Keywords: Multinational firms, FDI, Firm heterogeneity, Investment liberalization JEL Classification: F10, F23, L25, R12, R30

We acknowledge the financial support of the Institute of Developing Economies and JSPS Grant-in-Aid for Scientific Research (B). The authors would like to thank RIETI for research opportunities and the Ministry of Economy, Trade, and Industry (METI) for providing firm-level data. For useful comments, we thank Theresa Greaney, Taiju Kitano, Naoto Jinji, Toshiyuki Matsuura, Timothy Halliday, Yoshihiro Hashiguchi, Keiko Ito, Kozo Kiyota, Kaoru Nabeshima, and seminar participants at the Midwest International Trade Meeting, Asia Pacific Trade Seminar, Institute of Developing Economies, Japanese Economics Association Conference, Japanese International Economics Conference, Keio University, Kitakyushu University, and RIETI. The opinions expressed and arguments employed in this paper are the sole responsibility of the authors and do not necessarily reflect those of RIETI, METI, or any institution with which the authors are affiliated. All remaining errors are our own.

^{ζ} Corresponding author: Department of Economics, University of Hawaii at Manoa; e-mail: aritas@hawaii.edu

[§] Institute of Developing Economies; address: 3-2-2 Wakaba, Mihama-ku, Chiba-shi, Chiba 261-8545, Japan; e-mail: kiyoyasu tanaka@ide.go.jp

1. Introduction

Declining FDI barriers are dramatically restructuring the global economy. Advances in transportation and communication technology and multilateral reductions in FDI related policy barriers have led to a disproportionate expansion of production by multinational firms relative to less efficient domestic firms.¹ Today the total value-added generated by multinational foreign affiliates is over 10% of global GDP, more than twice the share contributed 20 years ago (UNCTAD, 1990 and 2011).² Given their superior technology and management expertise, multinational firms have been viewed as strong contributors to technological diffusion and economic development (Markusen, 2004). Nonetheless, their potential displacement effects on domestic activity have also generated growing anxiety towards their recent rise.

This paper seeks to quantify the impacts of FDI globalization on intra-industry reallocations and its implications for aggregate productivity. The firm-heterogeneity model of trade predicts that economic integration will lead to aggregate productivity gains and displacement of less productive firms. With the high fixed costs of foreign market participation, declining barriers favor the most productive firms in expanding abroad, whereas less productive firms contract under increased foreign competition (Melitz, 2003). Many previous studies have quantified the intra-industry reallocations resulting from trade liberalization (Pavcnik, 2002; Bernard et al., 2003).³ However, FDI plays a more dominant role in globalization and we know very little about the empirical magnitudes of reallocation being channeled through declining FDI barriers.⁴

We apply the simulation model developed by Eaton, Kortum, and Kramarz (2011, EKK

¹ While FDI barriers are not explicitly measurable, Gormsen (2011) estimates that FDI barriers have halved every 4.8 years and contributed to over 75% of the growth in FDI stocks in recent decades. ² Total value added (including demostic production) by multipationals empowered to give 25% of

 $^{^2}$ Total value added (including domestic production) by multinationals amounted to over 25% of global GDP.

³ Pavcnik (2002) finds that trade liberalization in Chile generated a 30% increase in aggregate productivity due to the reallocation of activity from less efficient to more efficient firms. Through ex-ante simulations of falling trade barriers, Bernard et al. (2003) quantified large reallocations of production activity from less to more productive firms among U.S. manufacturing firms. ⁴ Today worldwide sales of foreign affiliates are more than twice those of exports (UNCTAD,

⁴ Today worldwide sales of foreign affiliates are more than twice those of exports (UNCTAD, 2011).

hereafter) to Japanese multinational enterprises (MNEs) and estimate a model of heterogeneous firms facing fixed and variable FDI barriers. We use the general equilibrium model to simulate how multinationals respond to declining FDI barriers and quantitatively assess the resulting extent of intra-industry reallocations and its implications for aggregate productivity improvements. In an era of global competition, the welfare of small domestic firms is a widely covered concern, but excessive protection of relatively less productive firms could have adverse impacts on aggregate productivity growth. Given the politically sensitive nature surrounding FDI globalization, a quantitative assessment will be useful for policy discussions on FDI.⁵

We first provide evidence that the EKK model designed for trade is also consistent with Japanese multinational activity. Dissecting the micro-level data, we find supporting evidence for high fixed costs of multinational entry, costly technology transfer, Pareto distribution of foreign affiliate sales, and a weak pecking order of productivity sorting for multinational activities across markets. Estimating the structural parameters of heterogeneous firms by the simulated method of moments, we conduct model validation exercises. The results show that simulated multinationals fit the actual Japanese multinational activity quite well in terms of the moments used in estimation as well as out-of-sample activity that occurred under a significantly different environment of FDI barriers from the period used for estimation.

Having validated the model for data on Japanese multinationals, we use the estimated model as a quantitative tool to examine how multinational firms respond to declining investment barriers. Applying a hypothetical 20% reduction in both variable and fixed investment barriers, we find that falling barriers lead to asymmetric changes across heterogeneous firms at the intensive and extensive margins. Increased competition from inward FDI reduces domestic production by all firms and forces the least productive firms to exit. Due

⁵ The anxieties over globalization and the rise of multinationals are widely covered in the media. In 2008, 2 out of every 3 EU citizens believed globalization was beneficial only for large corporations while 78% of Americans viewed globalization favorably in 2002, which declined to 59% in 2007. After the 2008 financial crisis, the majority of Americans viewed it as a threat (Economist, February 19, 2009).

to high fixed setup costs of foreign production, the growth of foreign production is primarily yielded by the most productive firms. While the top quintile of firms grow by over 30%, the remaining bottom 80% of firms contract by over 3%. The resulting reallocation of production from less efficient to more efficient firms may generate up to a 30.7% improvement in aggregate productivity. Our numerical results are consistent with the actual asymmetric growth expansions observed over the last decade in Japan's manufacturing sector.

Our analysis is related to the literature that attempts to assess the gains from increasing multinational production. ⁶ Extending the neoclassical growth model to multinational production, Burstein and Monge-Naranjo (2009) and McGrattan and Prescot (2009) find that multinational firms generate welfare gains through "knowledge capital transfer". Using a Ricardian framework, Ramondo (2011) and Ramondo and Rodrigeuz-Claire (2009) find that openness to FDI leads to large gains through technology transfer. While these papers employ multi-country general equilibrium models to quantify welfare gains, their approach relies on aggregate measures of FDI activity and they do not explicitly account for the micro-level activities of individual firms and differences between fixed and variable FDI barriers.

By contrast, our application of the micro-simulation approach by EKK allows us to simulate firm-level changes from falling FDI barriers and quantify aggregate productivity gains through intra-industry reallocations. Since tracking the extent of market-share reallocation depends upon firm-level changes in sales and level of penetration across multiple foreign markets, it is important to model individual firm responses to FDI barriers. Combined with firm heterogeneity, fixed and variable barriers to FDI carry different consequences for firm-level conditions of entry and sales across markets because the overall magnitude of these impacts depends upon the distributional structure of firms and the sensitivity to these barriers. Our results show that the counterfactual changes in Japanese MNEs depend strongly on the nature of investment barriers; aggregate productivity improvements are larger from variable barriers

⁶ Our analysis is also related to recent micro-level studies on the location decisions of heterogeneous multinationals (Grossman et al., 2006; Aw and Lee, 2008; Yeaple, 2009; Chen and Moore, 2010).

compared to fixed barriers, as is consistent with the findings for exporting firms in di Giovanni and Levchenko (2010).

Before describing the details, we must emphasize that our application of the EKK framework captures many features of multinational production, but does not account for the role of trade.⁷ As previous work such as Helpman et al. (2004) shows that FDI and exporting decisions may be jointly related, our results must be interpreted with caution. If FDI and trade are substitutes, foreign activity by MNEs may displace part of their domestic activity. On the other hand, if FDI and trade are complementary, foreign operations may increase their domestic activity. However, existing evidence on the linkage between FDI and trade is mixed, making it difficult to predict the direction of bias in our simulations.⁸ In this study, we ignore the interaction effects between trade and FDI within firms, focusing solely on the market-share reallocations between firms.⁹

The rest of the paper is organized as follows. Section 2 describes our data sources and recent trends in the Japanese manufacturing sector. Section 3 presents an EKK model of multinational production, with supporting evidence from the Japanese data. Section 4 summarizes the procedure used to estimate structural parameters of the model and model-validation exercises. Section 5 presents counterfactual simulation analysis. Section 6 concludes.

⁷ This limitation is due to the lack of Japanese firm-level data on export sales by destination

⁸ Blonigen (2001) finds both complementary and substitution effects for the Japanese automobile multinationals investing in the U.S. while Yamashita and Fukao (2010) find that the overseas operations of Japanese MNEs do not reduce their domestic employment. Irarrazabal et al. (2010) find that with intra-firm trade, a reduction in FDI barriers could lead to increased domestic activity.

⁹ Recent studies such as Ramondo and Rodriguez-Claire (2009) and Garetto (2010) quantitatively examine FDI and exports in a unified framework by developing a structural model of traded inputs between multinational firms and their affiliates. However, these studies place less attention on the role of individual firm activity.

2. Data and the Stylized Facts of Japanese Multinational Firms

This section discusses the data sources for Japanese multinational and domestic firms. It then describes recent structural changes in the Japanese manufacturing sector to motivate our empirical study.

2.1 Data Sources

Data on Japanese firms come from the Basic Survey of Japanese Business Structure and Activities by the Japanese Ministry of Economy, Trade, and Industry (METI). The survey covers all business firms with 50 employees or more and capital of 30 million yen or more in both manufacturing and non-manufacturing sectors. Our analysis is based primarily on the 2006 survey. The 2006 sample accounted for 5% of total firms and 60% of total employees in the Japanese manufacturing sector.¹⁰ Meanwhile, data on foreign affiliate activity are from the Basic Survey of Overseas Business Activities by METI. The survey covers manufacturing and non-manufacturing in Japan and own at least one foreign business enterprise.¹¹ In the 2006 survey, there were 2,165 parent firms in manufacturing, which accounted for 31% of total employees of manufacturing firms in the Basic Survey of Japanese Business Structure and Activities.

After linking domestic parent firm activity to their manufacturing foreign affiliates and excluding the affiliates that were not operating and/or had no sales information, our dataset consists of 2,032 parent firms with 7,626 foreign affiliates across 70 markets in 2006. In this sample, the total sales by foreign affiliates were 99.1 trillion yen. The average Japanese multinational maintained four foreign affiliates with foreign sales of 5.7 billion yen per affiliate. However, domestic sales are missing for some parent firms, making it difficult to measure a linkage between domestic and foreign sales. Because including these parent firms does not alter

¹⁰ The surveyed sample included 12,855 manufacturing firms with 4.9 million employees in 2006.

¹¹ A foreign business is defined as one in which 10 percent of the affiliate's equity shares are owned by a Japanese parent firm.

our results quantitatively, we use the reduced sample of 1,656 parent firms with complete information on sales.

2.2 Structural Changes in the Japanese Manufacturing Sector

Dissecting the detailed firm-level data, we describe several important structural changes in the Japanese manufacturing sector that occurred during the past decade under a period of falling investment barriers. We first assign firms into percentile bins by their 1996 volume of global production which includes domestic, export, and foreign affiliate sales at the firm-level.¹² Table 1 tabulates the number of all firms and multinational parent firms for both years adjusting the sales intervals for the 1996 percentile bins into 2006 prices. Comparing these periods, we observe that the total number of firms declined by 8.9% from 14,117 to 12,855. On the other hand, the total number of multinational parents increased by 72.5% from 960 to 1,656. Decomposing the aggregate changes, we find that small firms disproportionately decreased in number, and large firms were more likely to become multinational. In terms of the extensive margin growth, foreign expansion was skewed towards larger firms.

=== Table 1 ===

Next, we examine production by Japanese firms. Between 1996 and 2006, the total volume of domestic production declined by 7.1%, whereas foreign production increased by 98.9%. The substantial growth of offshore production offsets the contraction in domestic production, whereby global production employing Japanese technology increased by 3.8% from 464.5 to 482.0 trillion yen. To illustrate the changes at the intensive margin, Table 2 shows the volume of production per firm in billions of 2006 yen across the percentile bins that are defined for 1996 and 2006, separately. In terms of the intensive margin, domestic production increased on average by 2.0%, and foreign production by 15.2%. On average, total production per firm increased by 14.0%.

¹² The following results are similar when the size is measured by domestic employment or when export sales are excluded.

=== Table 2 ===

Dissecting the aggregate changes reveals large differences across Japanese firms by size. In the bottom percentiles, firms experienced an increase in the average volume of their foreign production, but a fall in their domestic production. Together, their average volume of total production contracted. By contrast, firms in the middle and upper percentiles (30–99) increased their average production both at home and abroad. In the top 1%, a substantial increase in average offshore production dominated a fall in average domestic production, leading to a sizeable rise in total production per firm. Since these trends are drawn from cohort linked decile groups rather than panel linked firms, we cannot strictly interpret the changes occurring in the intensive margins. Nevertheless, the trends indicate that Japan's distributional production structure has substantially changed. Accounting for offshore production, global production per firm has increased disproportionately more for larger, multinational firms relative to smaller, domestic firms.

These findings have implications for aggregate productivity. As is shown in the literature on productivity dynamics, industry productivity growth can be attributed to reallocations of production shares of heterogeneous firms in that industry or within firm-level productivity improvements (Foster et al., 2001). Petrin et al. (2011) finds that micro-level reallocation of production resources from less to more productive plants generates a larger contribution to aggregate productivity growth for the manufacturing sector in the United States. However, prior studies on productivity dynamics have not necessarily distinguished between domestic and foreign production in measuring reallocation effects, and there has been little investigation of how foreign-market expansion may also improve aggregate productivity (Foster et al., 2001).

The observed reallocations in Table 2 are important because of the existence of firm heterogeneity in productivity, which is related to firm size and multinational status. Normalizing

value added per worker at the median value, we found that average labor productivity increases with the firm size for both 1996 and 2006 samples. In 1996, for instance, the smallest 10% of firms were on average 43% less productive than the average firm whereas the largest 10% of firms possessed a labor productivity advantage of over 160%. Furthermore, Tomiura (2007) finds that multinational firms in Japan are 60% more productive than non-multinationals. As large multinationals are more productive than smaller domestic firms, the observed market share reallocations over the past decade would necessarily translate into potentially large productivity gains without any technological improvements within firms.

For Japan, changes in *domestic* reallocation efficiency were found to account for over 80% of all aggregate productivity growth over the last decade (Inui et al., 2009). While Japan experienced a domestic productivity slowdown that has averaged less than 0.5% per year from 1994–2005 (Inui et al. 2009), we know little about how the reallocation of activity towards more efficient multinational production has impacted resource use and aggregate productivity. In this respect, the Japanese manufacturing industry offers an interesting case study for assessing productivity gains from reallocation of technological resources via multinational activity.

3. Theoretical Framework

Our theoretical framework is adapted from the heterogeneous exporting firm model of EKK (2011) which incorporates stochastic entry and demand stocks into a Melitz model. We adapt their model to multinational activity by allowing firms to engage in foreign markets through FDI, excluding the role of trade. Instead of incurring iceberg trade costs, heterogeneous firms must pay technology transfer and management costs to serve foreign consumers via offshore production. This section briefly presents a modified version of the EKK model for multinational production. Key elements of the model are monopolistic competition, Pareto distribution of firm efficiency, and variable and fixed costs of operating offshore production.

3.1 Firm Heterogeneity

There are *N* host countries with factor cost w_n . Each country has a continuum of potential producers, each producing a unique good *j* with efficiency $z_i(j)$. A set of firms that originate from country *i* can produce good *j* in country *n* with unit costs given by:

$$c_{ni}(j) = \frac{w_n d_{ni}}{z_i(j)} \tag{1}$$

Unit costs increase with host market's factor costs, w_n , and an iceberg form of efficiency loss d_{ni} , but decrease with firm-level productivity $z_i(j)$.¹³ Each pair of countries *i* and *n* is separated by technology barriers that rise in d_{ni} . In this setting, d_{ni} can be interpreted as the variable costs of operating offshore production that include information costs of working abroad, transaction costs of dealing with FDI policy barriers, or servicing network costs.¹⁴ We assume that a firm incurs no additional cost to implement its production technology at home, implying that there is no efficiency loss, $d_{ii} = 1$.

As in Melitz (2003), each firm receives a random productivity draw from a Pareto distribution. Then, a measure of potential producers with efficiency of at least z is:

$$\mu_i^Z(Z \ge z) = T_i z^{-\theta}, \quad z > 0 \tag{2}$$

 T_i indicates the average level of efficiency/technology in country *i*, and θ governs the distribution of firm-level productivities, which is assumed constant across countries. Since all goods are uniquely produced by a single firm and differ solely in the dimension of productivity, the terms "goods" and "firm" can be indexed interchangeably. As equation (1) leads to $z_i(j) = w_n d_{ni}/c_{ni}(j)$, the following equation describes the measure of goods that can be produced in country *n* by firms from country *i* with unit cost less than *c*:

¹³ The efficiency loss of offshore production is similar in interpretation to Keller and Yeaple (2008), showing that complex technologies deter multinational production from imports by multinational parent firms. It is also consistent with the findings in Ramondo (2011) and Ramondo and Rodriguez-Clare (2009) that multinational costs increase with distance, language barriers, and national borders. Also, intra-firm trade is an alternative explanation for variable costs in multinational production (Irarrazabal et al. 2010).

¹⁴ FDI policy restrictions may increase the costs of multinationals operating abroad in terms of ownership constraints, foreign-specific regulations, and weak legal protection of property and capital (World Bank Group, 2010).

$$\mu_{ni}(C \le c) = \mu_i^z \left(\frac{w_n d_{ni}}{c}\right). \tag{3}$$

Using equation (2), we can write equation (3) as $\mu_i^z \left(\frac{w_n d_{ni}}{c}\right) = T_i (w_n d_{ni})^{-\theta} (c)^{\theta}$.

3.2 Demand and Entry Costs

With constant-elasticity-of-substitution (CES) preferences, each country has the following demand function for differentiated products:

$$X_n(j) = \alpha_n(j) \left(\frac{p_n(j)}{P_n}\right)^{-(\sigma-1)} X_n \tag{4}$$

where $X_n(j)$ is the quantity sold by firm *j* in country *n*, X_n is an aggregate demand for manufacturing varieties, and P_n is the CES price index averaged across all goods consumed in country *n*. σ is the elasticity of substitution between any two goods with $\sigma > 1$, for which it is assumed $\theta - 1 > \sigma$. $\alpha_n(j)$ is an unobservable demand shock for firm *j* selling in country *n*, with higher values indicating a preferable shock to a variety produced by firm *j*.

A firm *j* enters market *n* by paying a fixed cost of setting up a production plant as follows:¹⁵

$$E_{ni}(j) = E_{ni}\varepsilon_n(j) \tag{5}$$

where E_{ni} is a general fixed cost that is constant for all firms entering market *n* from country *i*. The costs include the physical costs of building a plant and the information/FDI barrier costs associated with establishing a new affiliate in a new market. The information costs are related to marketing research, foreign contacts, and local recruitment of workers. FDI policy barriers in the form of fixed costs may include additional regulations required to set up an affiliate. $\varepsilon_n(j)$ is a idiosyncratic fixed cost specific to firm *j* entering market *n*, which accounts for unobservable factors of fixed costs. Higher values indicate larger investment costs for firm *j* to enter market *n*.

A firm *j* from country *i* operating in market *n* will generate net profits as:

¹⁵ EKK (2011) employ the marketing-based formulation of fixed costs in Arkolakis (2010) to account for an increasing fixed cost with the sales size. We first estimated such a form of entry condition, but found that the fixed entry cost is not quantitatively dependent on the sales size.

$$\pi_{ni}(j) = \left(1 - \frac{c_{ni}(j)}{p_n(j)}\right) \alpha_n(j) \left(\frac{p_n(j)}{P_n}\right)^{-(\sigma-1)} X_n - E_{ni}\varepsilon_n(j) \tag{6}$$

With monopolistic competition and Dixit-Stiglitz preferences, each firm maximizes its profit by charging a constant markup $\overline{m} = \sigma/(\sigma - 1)$ over its unit cost, $c_{ni}(j)$, such that $p_n(j) = \overline{m}c_{ni}(j)$. Then, its total gross profit is proportional to demand with a factor of $1/\sigma$, yielding $X_n(j)/\sigma$. Firm *j* will enter market *n* if and only if its operating profit is sufficient to overcome the fixed cost of entry:

$$\eta_n(j) \left(\frac{p_n(j)}{p_n}\right)^{-(\sigma-1)} \frac{X_n}{\sigma} \ge E_{ni} \tag{7}$$

where $\eta_n(j) = \alpha_n(j) / \varepsilon_n(j)$ is an entry shock to firm *j* that invests in market *n*.

3.3 Entry and Sales Conditions

The condition in equation (7) can be used to express the entry hurdle condition that firm j enters market n from country i if and only if its unit cost is less than the threshold entry cost:

$$c_{ni}(j) \le \bar{c}_{ni}(j) \tag{8}$$

where:

$$\bar{c}_{ni}(j) = \left(\eta_n(j) \frac{x_n}{\sigma E_{ni}}\right)^{1/(\sigma-1)} \frac{P_n}{\bar{m}}$$
(9)

Substituting the constant markup price and equation (9) into equation (4), we can express the latent sales conditional on entry:

$$X_{ni}(j) = \frac{\alpha_n(j)}{\eta_n(j)} \sigma E_{ni} \left(\frac{\bar{c}_{ni}(j)}{c_n(j)}\right)^{\sigma-1}$$
(10)

Equations (8), (9), and (10) provide the main theoretical predictions about the structure of heterogeneous multinational firms with varying efficiency. The entry hurdle in (9), $\bar{c}_{ni}(j)$, pins down the critical threshold for entry, whereby a lower $\bar{c}_{ni}(j)$ indicates that the market is less attractive for firms to engage in multinational production. As unit cost decreases with firm-level productivity, more efficient firms are more likely than less efficient firms to overcome the entry hurdle. Conditional on entry, equation (10) dictates the volume of sales generated by firm *j* in

that market.

Analogous to the behavior of exporting firms found in EKK, the model then predicts that higher productivity firms are more likely than less productive firms to: (i) invest in a larger number of markets, (ii) penetrate the less attractive markets, and (iii) yield larger sales per market. With varying entry hurdles, firms first enter the most attractive market, and then invest progressively in less attractive markets, predicting a hierarchy of market destinations in which more productive firms progressively enter less attractive markets. However, the presence of entry and demand shocks in the model allows firms with identical productivity to deviate from identical patterns of market entry and volume of affiliate sales. Thus, the model predicts a weak pecking order.

3.4 Empirical Regularities of Japanese Multinational Production

Since we apply the EKK model of trade to FDI, it is necessary to examine whether empirical regularities of multinational activities are consistent with the model, as is found in exporting firm data. Using Japanese firm-level data of multinational entry and sales across foreign markets, we document a number of the stylized patterns in line with the theoretical implications of the EKK model.

Market Entry and Market Size

Figure 1 shows a relationship between market entry and market size. Panel A plots the number of parent firms investing in each market against the market size, as measured in real GDP in billions of 2000 U.S. dollars.¹⁶ The number of firms investing in each market increases with the size of the host market. Panel B shows the average sales of foreign affiliates in each market against the market size. Sales per affiliate increase with the market size, as is consistent with the number of investing parent firms. Thus, larger markets may attract the entry of more

¹⁶ Data on real GDP come from the World Development Indicators.

multinational firms, but have a weaker positive impact on average affiliate sales. The patterns observed in Panel A and Panel B can be well explained by the existence of fixed entry costs.

=== Figure 1 ===

Sales Distributions

Figure 2 plots the sales distributions of Japanese firms within individual markets. For the Japanese market, firm size is measured by domestic production that aggregates firms' sales in Japan and their export markets. We normalize each firm's sales relative to the mean of domestic sales and compute a fraction of firms with at least that much sales.¹⁷ For other foreign markets, we use the total sales of foreign affiliates, including their sales to local, home, and third markets. Figure 2 shows a plot of the normalized sales and the fraction of firms for Japan, China, the U.S., and Thailand. The shapes of the sales distributions are similar across markets and closely following a linear line in the upper tail, supporting the assumption that underlying firm-level productivity of multinational sales is Pareto distributed. However, the distribution starts to deviate from a linear line in the lower tail. As argued by EKK, the deviations in the lower tail can be partly explained by market-entry shocks.

=== Figure 2 ===

Market Entry and Sales in Japan

Figure 3 describes the relationship between market entry and sales in Japan. We group firms into different sets according to their entry to foreign markets. First, Japanese firms are sorted into the set of firms with the *minimum* number of foreign markets they penetrated. Panel A in Figure 3 plots the average sales in Japan for each set of firms investing in at least m markets or more. Sales in Japan rise monotonically with the number of markets where multinationals own an affiliate. Second, firms are grouped by the number of markets they

¹⁷ The fraction is constructed by dividing (rank – 0.5) by the total number of firms/affiliates for each observation in a given market, where the rank is one for a firm/affiliate with the largest sales.

entered. Panel B presents a plot of average sales in Japan against the number of firms investing in *m* markets or more, with the marker indicating the number of markets they served. For the set of firms penetrating a single market, there are over 1000 firms with relatively low average sales in Japan. As the number of markets served by them increases, the set of these firms becomes smaller with higher levels of sales. Figure 3 identifies a significant link between domestic sales and market entry, indicating that larger firms are more likely than smaller firms to invest in a larger number of markets. This pattern is consistent with the productivity sorting pattern found in heterogeneous firm exporting decisions across multiple markets, highlighting the critical role of productivity in self-selection decisions of FDI activity.

=== Figure 3 ===

Multinational Production Intensity

Lastly, Figure 4 describes the relationship between sales in Japan and foreign affiliate sales. We compute the normalized sales of foreign affiliates in each market defined by $(X_{nJ}(j)/\overline{X}_{nJ})/((X_J(j)/\overline{X}_J))$. To remove the market effect, firm *j*'s sales in market *n* are divided by the average sales in that market, and then are further divided by the normalized sales in market J, Japan, to remove the firm-size effect. The figure plots the 90th percentile and median normalized sales of foreign affiliates against the number of parent firms for each market. Normalized affiliate sales increase with the popularity of foreign markets as measured by the number of firm entry. However, the pattern is noisier for the markets in which less than 10 firms entered. The generally positive relationship suggests that fixed costs alone cannot explain the varying levels of multinational participation and point to the existence of variable FDI costs such as costly technology transfer.

=== Figure 4 ===

4. Estimation and Model Validation

In this section we estimate the structural parameters of the model and evaluate the model's ability to replicate real multinational activity. Before presenting the results, we briefly review the estimation strategy and simulation procedure for individual multinational firms. Because we closely follow the framework developed by EKK (2011), we describe details of the estimation framework in Appendix.

4.1 Respecification for Simulation

Following EKK (2011), we re-specify the entry and sales conditions in equations (8), (9), and (10) for quantification. We first isolate the heterogeneous component of unit costs by defining $u(s) = T_J z_J(s)^{-\theta}$ as the standardized unit costs for a Japanese firm *J*. Redefining the efficiency of potential producer in Japan in terms of standardized unit cost for firm *j*, we arrive at:

$$u(j) = T_I z_I(j)^{-\theta} \tag{11}$$

where *J* denotes Japan as a home country. Second, the country-level parameters in equations (9) and (10) are connected with the total number of foreign entries N_{nJ} and FDI shares X_{nJ}/X_n , respectively. We can then express the entry hurdle as:

$$u(j) \le \bar{u}_{nJ}(\eta_n(j)) = \kappa_2 N_{nJ} \eta_n(j)^{\hat{\theta}}$$
(12)

where

$$\begin{split} \tilde{\theta} &= \frac{\theta}{\sigma - 1} > 1 \\ \kappa_2 &= \int \eta^{\tilde{\theta}} g_2(\eta) d\eta \end{split}$$

Note that $\bar{u}_n(\eta_n(j))$ is a standardized entry hurdle in market *n* for potential producer *j* from Japan. $\tilde{\theta}$ is observed heterogeneity in sales, with a lower value indicating a larger dispersion in sales. Conditional on entry, the sales condition for firm *j* in market *n* is given by:

$$X_{nJ}(j) = \frac{\alpha_n(j)}{\eta_n(j)} \overline{X}_{nJ} \frac{\kappa_2}{\kappa_1} \left(v_{nJ}(j) \right)^{-1/\widetilde{\theta}}$$
(13)

where:

$$\kappa_{1} = \kappa_{0} \iint \alpha_{n}(j)\eta_{n}(j)^{(\bar{\theta}-1)} g(\alpha,\eta) d\alpha d\eta$$
$$\kappa_{0} = \frac{\tilde{\theta}}{\tilde{\theta}-1}^{.18}$$

 $v_{nJ}(j) = \frac{u(j)}{\overline{u}_{nJ}(\eta_n(j))}$ follows a uniform distribution on [0, 1].

To parameterize κ_1 and κ_2 , $g(\alpha, \eta)$ is assumed to be joint lognormal, with variances (σ_{α} and σ_{η}), and correlation ρ . The joint lognormal distribution allows us to write κ_1 and κ_2 as follows:

$$\kappa_{1} = \left[\frac{\tilde{\theta}}{\tilde{\theta}-1}\right] exp\left[\frac{\sigma_{\alpha}+2\rho\sigma_{\alpha}\sigma_{\eta}(\tilde{\theta}-1)+\sigma_{\eta}(\tilde{\theta}-1)^{2}}{2}\right]$$
(14)

$$\kappa_2 = exp\left[\frac{\left(\tilde{\theta}\sigma_\eta\right)^2}{2}\right] \tag{15}$$

4.2 Estimation Procedure

Taken together, the entry and sales conditions are governed by four structural parameters: heterogeneity in observed sales $\tilde{\theta}$, variance in sales σ_{α} , variance in entry shocks σ_{η} , and correlation ρ . We denote the set of these structural parameters by:

$$\Theta = (\tilde{\theta}, \sigma_{\alpha}, \sigma_{\eta}, \rho)$$

Comparing moments of the simulated and actual datasets on multinational firms, we search for an optimal set of structural parameters via the simulated method of moments.

In the first step of this estimation procedure, we use the entry and sales conditions in equations (12) and (13) to simulate individual multinational activities. An artificial producer, s, is generated by its efficiency draw, u(s), sales shock, $\alpha_n(s)$, and entry shock $\eta_n(s)$. With an initial guess for the structural parameters and aggregate data, we can produce a dataset of artificial multinational firms with varying productivity and the location and sales of their foreign

 $^{^{18}}$ Refer to the derivation of the price index in the Appendix A.

affiliates.

In the second step, we construct a set of moment conditions from simulated multinationals and actual Japanese multinationals. We define a vector of deviations between actual and artificial moments for outcome k:

$$y(\Theta) = m^k - \hat{m}^k(\Theta). \tag{17}$$

Following the theoretical implications of the model, we choose four moment conditions: pecking order strings, affiliate sales distributions across markets, parent sales distribution in Japan, and multinational production intensity.

We seek to capture these patterns that were identified in the empirical regularities of Japanese firms. Specifically, the pecking order strings are defined as a group of multinational firms entering each possible combination of the five most popular countries in terms of multinational entry: China, United States, Thailand, Taiwan, and Indonesia in 2006. Next, the distribution of affiliate sales is a group of multinationals investing in each foreign market and belonging to the actual sales percentiles for $q = 50^{\text{th}}$, 75^{th} , 100^{th} . Then, the distribution of multinational parent sales in Japan is a group of multinationals entering each foreign market and belonging to the percentiles of actual sales in Japan for $q = 50^{\text{th}}$, 75^{th} , 100^{th} . Finally, the multinational production intensity is a group of multinationals whose ratio of sales in market *n* to sales in Japan is below and above the 50^{th} percentile for each market.

Finally, we estimate a set of structural parameters that match the simulated moments as closely as possible with moments from the actual data. Stacking a vector of moment conditions, we minimize the objective function:

$$\widehat{\Theta} = \arg\min_{\Theta} \{ [m^k - \widehat{m}^k(\Theta)]' \ [m^k - \widehat{m}^k(\Theta)] \}.$$
(18)

4.3 Estimation Results

Table 3 presents the estimation results of the structural parameters with bootstrapped standard errors. In column (1), we report the benchmark results. To mitigate the chance that

noisier segments of the data adversely influence the estimates, we exclude markets with less than 10 foreign affiliates from the estimation. We find that the key distribution parameter of size dispersion, $\tilde{\theta} = \theta/(\sigma - 1)$, is 1.9. This estimate is slightly lower than EKK's estimate of 2.46 for French exporting firms, indicating that Japanese multinationals may have a greater dispersion in sales than French exporting firms do. Given that our estimates are generated from multinational, not exporting, activities, the parameter estimates are not strictly comparable. Nevertheless, it is reassuring that our estimate for firm heterogeneity is quantitatively similar to the estimate of French exporters found by EKK.¹⁹

We find that the estimate of variance in sales shock is 1.64, which is quite similar in magnitude to the corresponding estimate for French exporters. The estimate of variance in entry shock is 0.39, which is also quantitatively close to the corresponding estimate of 0.34 for French exporters. While there is no clear reference to interpret the size of these estimates, these results indicate that heterogeneity in productivity is not sufficient to account for the entry and sales patterns of actual Japanese multinationals. Comparing the variances of entry and sales shocks, we find that the model is more effective at explaining variations in market entry than variations in foreign affiliate sales. Lastly, we find that the correlation between entry and sales shocks is -0.62, suggesting that the entry shocks are negatively associated with the sales shocks.

=== Table 3 ===

We proceed to check the robustness of the benchmark estimates. First, a possible concern is that the exclusion of smaller markets could influence the benchmark estimates. Estimating the parameters for all the markets, we present the results in column (2). The results show that the estimate for size dispersion rises to 2.12, but the overall point estimate and bootstrapped standard errors are quite similar to the benchmark. Second, we check the sensitivity of the

¹⁹ The level of size dispersion is related to heterogeneity in firm efficiency via an elasticity of substitution. As Kang (2008) estimates that the elasticity of substation is 2.19 for the Japanese manufacturing sector in 1994-2004, a rough estimate of efficiency dispersion from our estimated parameter is 2.37. By contrast, Wakasugi et al. (2008) find that the dispersion parameter of total factor productivity was 1.69 for Japanese manufacturing firms in 2003.

benchmark parameters to a different set of moments. Among the moments used, the pecking order of entry may not fit as well for multinational production compared to export entry because FDI activity is encouraged by a multitude of investment motives, including market access, efficiency seeking, and policy incentives. To address this concern, we exclude the pecking order moments from the estimation. Column (3) shows that the estimated parameters and bootstrapped standard errors are similar to the benchmark results, suggesting that our results are robust to the different set of moments used.

Finally, we estimate the structural parameters for Japanese multinational activity in 1996. Column (4) shows that the estimate of size dispersion rises to 2.13, but the difference falls within the range of the bootstrapped standard errors from the benchmark result. While the other estimates differ slightly from the benchmark estimates, there is no substantial deviation. Taken together, our robustness checks demonstrate that the benchmark estimates of the four parameters are not sensitive to alternative specifications of the sample and the moments used for estimation.

4.4 Model Validation

Having checked the robustness of key structural parameters, we proceed to examine how well the estimated model performs in replicating real multinational activity. First, we assess the model's ability to replicate the actual data as captured by our moment conditions. Given the estimated parameters, we simulate multinational activity and compare the simulated moments with the moments from the estimation sample. In results not shown, we find a good fit of the data between simulated and actual moments, suggesting that the model is able to closely replicate the in-sample moments of the actual data.²⁰ As the same sample is used for both estimation and validation, the exercise is a conventional internal model validation test.

Since in-sample replications are based only on the information conditioned in the estimation of structural parameters, the internal validation test does not clearly show the

²⁰ Refer to section 6 in Arita and Tanaka (2011).

model's ability to predict multinational activity in an environment with a significantly different level of FDI barriers. To gain further confidence that the model can be used to simulate multinational activity when FDI barriers change, we examine the external validity of the model. We make use of data from a decade earlier on Japanese multinationals, and investigate how well the model is able to reproduce out-of-sample predictions of Japanese multinational activities in 1996 with our parameters estimated on 2006 decade. Using 1996 data to parameterize N_{nJ} and \overline{X}_{nJ} , we simulate an artificial set of multinationals from the entry and sales conditions:

$$u(s) \le \bar{u}_n(\eta_n(s)) = N_{nJ}^{1996} \kappa_2 \eta_n(s)^{\tilde{\theta}}$$
⁽¹⁹⁾

$$X_{nJ}(s) = \overline{X}_{nJ}^{1996} \frac{\alpha_n(s)}{\eta_n(s)} \frac{\kappa_2}{\kappa_1} \left(\frac{u(s)}{\overline{u}_n(s)}\right)^{-1/\widetilde{\theta}}.$$
 (20)

Because FDI barriers were likely to change significantly from 1996 to 2006, this external validation approach is in the spirit of the "non-random holdout sample" (Keane and Wolpin, 2007). That is, we assess a model fit by asking how well the model can replicate multinational activity outside the support of the data along the dimension that the model is meant to predict, i.e., changes in FDI barriers.

Figures 5.1 and 5.2 present the out-of-sample simulations for 1996 activity. The figures are scatter plots of the number of simulated firms vs. the actual number of firms according to the bins defined by the moment conditions estimated in the model. Panel A presents the number of firms falling into each of the 32 pecking order strings for the real and simulated data. Panels B through D present the number of firms falling into the defined percentile groups across foreign markets of each respective moment. A 45-degree straight line is plotted as a reference to indicate a perfect fit between simulated and actual firms.

From Panels A through D, we find that the model fit is considerably strong along various dimensions of multinational activities. The good fit is evident from the four moments: pecking order strings, distribution of foreign affiliate sales across markets, distribution of sales back in Japan, and multinational production intensities. However, we tend to slightly under-simulate

sales of foreign affiliates and domestic sales in Japan.²¹ Although the out-of-sample predictions reveal slight deviations from the data, the overall simulations are externally consistent with the Japanese multinational activity a decade ago.²² This gives us some degree of confidence that the model can be used to simulate multinational activity under reduced FDI barriers.

=== Figures 5.1 and 5.2 ===

5. Counterfactual Analysis

Internal and external model validation provides a degree of confidence that the model can be used to simulate multinational activity under falling FDI barriers. In this section, we use the model to perform counterfactual simulations to investigate the impacts of changes in FDI barriers on intra-industry reallocations and the implications for aggregate productivity.

5.1 Global General Equilibrium

For counterfactuals, we first need to account for adjustments of aggregate prices and wages that take place following an exogenous change in variable and fixed FDI barriers. Leaving details of the general equilibrium setting and data sources in Appendix D, we briefly summarize the counterfactual simulation procedure.

Following the approach in EKK (2011), we first set up a general equilibrium framework in which producers serve their home and host countries through FDI. Each country is endowed with labor, which is mobile within countries, but immobile across countries. Intermediates are a Cobb-Douglas combination of labor and intermediates. Final output is non-traded and a Cobb-Douglas combination of manufactures and labor. Fixed cost for FDI is paid by labor.

²¹ In comparison with in-sample predictions for 2006 activity, the predictive ability of the model appears to be slightly weaker in out-of-sample predictions for 1996 activity (Arita and Tanaka, 2011).

²² We also found some shortcomings of the model. First, large firms tend to invest in too many countries, but small firms invest in too few. Second, we tend to under-simulate the sales level of Japanese multinationals in both home and foreign markets. Lastly, the model poorly explains vertically motivated multinationals.

Profits accrue to the headquarter country of producers. As consumers own equal shares of each firm headquartered in their country, the profits are redistributed equally among the consumers. A country's GDP is equal to its total wage from production in its own country and its total profit from abroad. Lastly, some countries are net receivers for FDI, and we allow for FDI deficits.²³

The general equilibrium framework is set up such that manufacturing production and consumption across countries are connected through FDI activity. Equilibrium in the world market for manufacturers leads to a system of equations. Based on the approach of Dekle, et al. (2008), changes in wages and prices can be solved as a result of an exogenous change in variable and fixed FDI costs. By solving changes in prices and wages jointly, we can calculate counterfactual changes across countries in FDI sales from Japan and the entry number of Japanese firms, \hat{X}_{nI}^{c} and \hat{N}_{nI}^{c} .

Given the counterfactual changes in aggregate sales and entry from Japan to other countries, we can calculate the expected changes in the aggregate sales and entry. Then, we use the entry and sales conditions in equations (12) and (13) to specify the corresponding counterfactual conditions:

$$u(s) \le \bar{u}_{nJ}^{\mathcal{C}}(\eta_n(s)) = N_{nJ}^{\mathcal{C}} \frac{\eta_n(s)^{\tilde{\theta}}}{\kappa_2}$$
(21)

$$X_{nJ}^{C}(s) = \overline{X}_{nJ}^{C}(s) \frac{\alpha_{n}(j)}{\eta_{n}(j)} \frac{\kappa_{2}}{\kappa_{1}} \left(\frac{u(s)}{\overline{u}_{n}^{C}(s)}\right)^{-1/\widetilde{\theta}}$$
(22)

Holding the structural parameters fixed, we next simulate a set of artificial firms on the basis of equations (21) and (22), generating a counterfactual dataset of multinational entry and sales. For our counterfactual simulations, we fix productivity draws and entry and sales shocks specific to individual firms in order to ensure that all changes in firm-level activity relative to the baseline stem solely from a change in FDI barriers.

For our analysis, a hypothetical experiment should provide a useful insight into the response of individual firms to aggregate shocks. However, it should be emphasized that the model is

²³ FDI deficits are fixed to their 2006 levels in counterfactuals.

based on a set of highly stylized assumptions. As emphasized in counterfactual exercises by Eaton and Kortum (2002) and Chor (2010), counterfactuals should not be interpreted definitely, but should be viewed as an explorative investigation.

5.2 Global Reductions in FDI Barriers

In our first experiment, we examine the impacts of a 20% global reduction in both variable and fixed FDI barriers by applying an exogenous change in the variable costs, $\hat{d}_{ni} = 1/1.20$, and that in the fixed costs, $\hat{E}_{ni} = 1/1.20$, for all $n \neq i$. These reductions can be interpreted as a decline in the variable and fixed costs of manufacturing production abroad, which result from advances in communication and transportation technology and continued multilateral liberalization of FDI policy restrictions. Production barriers within a country are assumed to remain unchanged by setting $\hat{d}_{ii} = 1$ and $\hat{E}_{ii} = 1$. We first present counterfactual changes at the aggregate level and then the changes that occur at the firm level.

Column (1) in Table 4.1 presents the general equilibrium changes for real wages at the aggregate level. We find that falling FDI barriers lead to a general increase in real wages around the world (approximately 29%). This result is generated by two forces. First, foreign affiliate expansion by multinationals leads to a rise in income around the world due to the extra income that their headquarters country receives from higher profits generated from multinational activity abroad. Second, the increased production reduces the aggregate price level because more efficient multinational firms produce at lower marginal costs. Specifically, the countries that are more open to FDI such as Hong Kong, Singapore, and Belgium, would experience a larger increase in inward multinational activity, and receive greater welfare gains in terms of a steeper decline in prices and a more dramatic increase in real wages. On the other hand, the countries that have higher levels of outward FDI such as Japan, the United States, and Netherlands, may experience a modest increase in prices because of the increased wealth effects from multinational profits generated abroad. Thus, they receive smaller welfare gains in terms of

a modest decline in prices and a modest increase in wages. In the case of Japan, small inward FDI and large outward FDI combine to yield a modest 0.7% increase in real wages.

===Table 4.1===

With the general equilibrium changes in wages and prices, we can predict aggregate changes in entry and sales by Japanese firms across foreign markets and in Japan. Columns (2) and (3) in Table 4.1 show that Japanese firms increase their entry and sales in the majority of foreign markets. On average, the entry of Japanese firms increases by 72.5% and their sales rise by 103.4%. However, these changes differ significantly across markets. As falling inward barriers raise competition in the Japanese home market through lower aggregate prices, Japanese firms decrease in number by 2.7% and reduce their aggregate domestic production by 3.5%. By contrast, falling outward barriers lead to foreign expansion by Japanese firms.

Table 4.2 presents the results for the aggregate number and production of Japanese firms according to these aggregate changes around the world. Multinational firms increase in number by 66.4% and expand their aggregate foreign production by 107.3%. Because foreign expansion dominates domestic contraction, the total volume of domestic and foreign production increases by 18.9%. Thus, our simulations show that falling FDI barriers abroad lead to a significant restructuring of the economy from domestic to multinational production.

===Tables 4.2, 4.3, and 4.4===

Simulating individual firm activity, we then dissect how falling FDI barriers affect firms differently according to productivity levels. Table 4.3 shows the counterfactual results for changes in the number of total firms and multinational firms, i.e., the extensive margin, across productivity decile groups. Due to high fixed entry costs in foreign markets, most firms in the lowest decile group are domestic whereas most multinational firms belong to the higher productivity groups. Falling investment barriers induce 1,004 domestic firms to become multinational: 176 from firms below the 50th percentile in terms of productivity and 828 firms with productivity above the 50th percentile. There is no growth at the extensive margin for the

top 1% firms as they all are already multinational in the baseline. Furthermore, the increased competition from abroad leads to an exit of 350 domestic firms, which are concentrated in the bottom 10% decile group with a net exit rate of 25%.

These asymmetric changes along the extensive margin highlight the important role of fixed costs in multinational production and the self-selection decision to enter foreign markets. When fixed barriers fall, the impact at the extensive-margin is pronounced for those firms closest to the threshold productivity prior to the change. If fixed costs are sufficiently high, falling barriers will more likely provide foreign production opportunities for higher productivity firms. In addition, among potential producers below the cutoff productivity, higher productivity firms tend to overcome entry barriers across a larger number of markets.

Table 4.4 presents the changes in domestic and total production per firm and foreign production per multinational, i.e., the intensive margin. Falling inward barriers lead to the entry of foreign firms to the domestic market, and increased competition at home leads to an across-the-board reduction in the average domestic production. In contrast, falling outward barriers lead to a rise in foreign production per multinational firm across all firms, whereby intensive margin growth is significantly higher for higher productivity firms. Firms below the 70th percentile grow by less than 2 billion yen (or 3% of their total production), while the top 1% grow by 486.6 billion yen (or 57% of their total production). Because the intensive margin of foreign productivity firms decreases by 0.9 - 2.5%. In contrast, total production per firm increases for firms in the upper percentile groups, with an extremely large expansion for the top 1% of firms by 481.3 trillion or 56.3% of their total sales.

The decrease in variable costs yields asymmetric impacts along the intensive margin. A percentage decline in variable costs translates into a disproportionate increase in the volume of sales for firms that have larger foreign production across markets. For firms below the 70th percentile that generate less than 0.1% of all their sales abroad, the impacts of falling barriers on

total production growth are small, contributing to a less than 0.9% increase in total sales. This contrasts with the firms above the 70th percentile that generate more than 50% of their sales abroad; falling barriers contribute to a more than 30% increase in their overall sales. These more productive firms penetrate a larger number of markets and yield greater sales per market, magnifying the overall scale effect.

The asymmetric changes at the extensive and intensive margins lead to significant intra-industry reallocation of market shares across firms. By calculating the percentage share of production according to quintile productivity groups, we depict the market shares in baseline and counterfactual simulations in Figure 6. Prior to any reduction in investment barriers, the baseline production structure is already heavily skewed—the top quintile of firms account for 56.6% of production. Following a 20% reduction in variable and fixed FDI barriers, the market share of this group increases to 64.5% while the remaining bottom firms lose market share.

=== Figure 6===

This reallocation from less to more productive firms may lead to large aggregate productivity gains. To quantify productivity improvements, we adopt the productivity decomposition approach of Foster et al. (2001). Specifically, we calculate the following components of aggregate productivity changes:

$$\Sigma_s \ln(z_s) \Delta W eight_s \tag{23}$$

$$\sum_{s} Weight_{s} \left[\ln \left(z_{exit_{s}} \right) - \overline{\ln \left(z_{s} \right)} \right) \right]$$
(24)

where *Weight_s* is the output share of simulated firm *s* in the total production of all firms, z_s is a measure of the simulated firm's efficiency, and $\overline{\ln(z_s)}$ is the average efficiency level of the industry. Compared with the standard decomposition technique, our model does not allow for within-firm technology improvement. If we assume that firm productivity is identified solely by its efficiency draw *z*, then, the above two measures can be used to assess aggregate productivity changes stemming from the intra-industry reallocations. We find that a 20% reduction in FDI barriers translates into an approximate 29.6% gain from market-share reallocations and a 1.2%

gain from the exit effects. Together, aggregate productivity improves by 30.7%.

It is important to note that these aggregate productivity gains are primarily realized from expanding multinational production in Japan's investment partners, rather than in its domestic industry. Thus, it is more appropriate to attribute our measured productivity improvements to reallocations of Japanese technological resources abroad, of which gains are realized at the global, rather than domestic, level.

5.3 Variable vs. Fixed FDI Barriers

We now examine more closely the different consequences between reductions in variable and fixed FDI barriers on intra-industry reallocations and aggregate productivity changes. For this task, we run counterfactual simulations for (1) a 20% global decline in strictly variable FDI barriers, and (2) a 20% global decline in strictly fixed FDI barriers. These comparisons allow us to explicitly examine the impacts between variable and fixed FDI barriers.

Before presenting the results, we clarify the distinction between variable and fixed costs for FDI activity. Variable costs are associated with technology/management efficiency losses, transaction costs, and policy factors such as foreign taxes. The model predicts that falling variable costs reduce the unit cost of foreign production for each firm, which increases the probability for firms to enter a market and expands their production there. On the other hand, fixed costs capture the establishment costs of offshore production, including plant construction, information, and network costs to set up a foreign affiliate. A decline in the fixed costs encourages a firm to overcome the entry hurdle in a market. Beyond any secondary effects on the price level, falling fixed barriers are related to the intensive margin of multinational sales through the sales condition.

Table 5.1 reports the aggregate outcomes from reducing variable and fixed barriers, separately. The total number of firms declines by 2.21% for the variable barriers and by 1.53% for the fixed barriers, suggesting that variable costs have a larger impact. In contrast, fixed costs

have a slightly higher impact on multinational firms than variable costs; multinationals increase in number by 28.6% for fixed costs and by 25.2% for variable costs. This suggests that fixed costs play an important role in accounting for the entry of multinationals. In terms of the volume of production, the variable costs have a significantly stronger impact than the fixed cost. Foreign production increases by 59.1% for the former and by 27.2% for the latter. Reductions in variable barriers leads to an increase in foreign competition, whereby domestic production contracts by 0.74% for the variable barriers. In the case of reducing fixed barriers, the wealth effects from increased Japanese multinational profits generated abroad dominate the pro-competitive effects of increased inward foreign multinational production, leading to an increase in domestic production by 0.28%. Together, the expansion of total production is larger from falling variable barriers.

=== Tables 5.1, 5.2, and 5.3===

Table 5.2 reports the counterfactual changes along the extensive margin across productivity decile groups. We find that reduced fixed barriers induce 455 additional firms to become multinational, while reduced variable barriers induce 415 additional firms to invest abroad. This suggests that fixed entry costs play a larger role in the extensive margin of FDI. Once again these impacts are more pronounced for higher productivity firms than lower productivity firms.

Table 5.3 reports the changes along the intensive margin. In terms of foreign production, we find that falling variable barriers induce the top 1% of firms to increase their intensive margin by 314.8 billion yen, and by 82.5 billion yen in the case of fixed barriers. Thus, relative to reductions in fixed barriers, reductions in variable barriers yield more dramatic growth along the intensive margin of multinational production.²⁴ Overall we find a larger reallocation of total production for variable barriers relative to changes in fixed barriers.

²⁴ Reductions in variable inward barriers lead to a reduction in average domestic production across all firms except for the lowest decile group which increases because of exiting by the smallest firms. While all firms decrease domestic production, due to the disproportionate exiting of small firms, there is an increase in average domestic production of all firms. With wealth effects, the reductions in fixed barriers lead to an increase in domestic production for all firms.

As variable investment barriers fall, the most productive firms expand their market share progressively, which in turn yields an increasing level of reallocation effects. By contrast, an increase in productivity gains is modest for declining fixed barriers, which have a stronger impact on the extensive margin than the intensive margin. Because changes along the extensive-margin generate a weak expansionary effect on the incumbent multinational firms, there would be a small contribution of the reallocation effects to aggregate productivity improvements. To clarify this point, Figure 7 shows the counterfactual changes in market shares from falling variable and fixed barriers as compared with the baseline. It is evident that variable barriers lead to a larger reallocation of market share from less to more productive firms.

=== Figure 7 and 8===

Assessing the impact of these reallocations on aggregate productivity, we find that falling variable barriers lead to an approximate 17.3% improvement in aggregate productivity. By contrast, falling fixed costs lead to a much smaller improvement of 4.5%. Figure 8 plots the simulated gains in aggregate productivity from varying the level of changes in variable and fixed levels of FDI barriers. The simulations indicate that aggregate productivity improvements are driven by reductions in variable FDI barriers as is in line with the findings for exporting firms in di Giovanni and Levchenko (2010). Previous studies on gains from FDI have not explicitly distinguished between these types of barriers. Our numerical results demonstrate the importance of modeling each type of barriers as they carry substantially different quantitative implications for aggregate productivity improvements.

5.3 Discussion

Our simulations present the first attempt to explicitly quantify the intra-industry reallocations occurring from FDI globalization. This raises a question about the quantitative relevance of our results. To assess the contribution of FDI globalization to aggregate productivity growth, we relate counterfactual simulations to the actual structural changes that occurred in Japan's

manufacturing sector over the last decade (1996–2006). Given that the model simulations are based exclusively on reductions in FDI barriers and cannot account for all other economic factors that affected Japan between 1996 and 2006, such a comparison is limited. Nonetheless, this period of falling FDI barriers serves as a useful benchmark to gauge the quantitative magnitude on various changes in FDI barriers and to assess what dimensions of the actual historical changes are captured well in our simulations.

=== Table 6===

Table 6 compares counterfactual simulations generated from 25%, 20%, and 10% reductions in variable and fixed barriers against the actual changes that occurred over the last decade. We find that the counterfactual simulations are able to capture several important dimensions of the actual historical changes. First, a 20% reduction in FDI barriers delivers a similar magnitude of multinational growth that occurred within the last decade, fairly capturing the growth in foreign production abroad (106.2% vs. the actual 99.0% growth) and the growth in the number of multinational firms (66.7% vs. the actual 72.0% growth).

Second, the simulations roughly follow the asymmetric growth pattern experienced in Japan. In the actual changes, large expansions in multinational activity and contractions in domestic activity led to higher realized growth for the top productivity quintile (14.7%) relative to the lower 80% of firms (6.5%). The model generates similar patterns of high growth for the top firms (7–36%) relative to the less productive firms. Unlike the actual changes, our simulations find a contraction of the bottom 80% of firms. This difference is expected given that our model is based on static simulations that do not account for any growth aspects that occurred during the ten year period. Importantly, the intra-industry reallocations theorized by the model are strongly supported by both the data and simulations: larger, more productive firms expand at a disproportionate rate relative to smaller, less productive firms in both the real and simulated realizations of data.

Overall, a 15–25% reduction in variable and fixed FDI barriers delivers substantial reallocation of Japanese technological resources, which loosely follows the structural changes

that occurred over the last decade. This level of reduction roughly translates into aggregate productivity improvements in the range of 10–40%. To place this in perspective, over the last decade, Japan's domestic aggregate productivity grew by less than 10%.²⁵ Thus our numerical results suggest that the disproportionate expansion of highly productive multinationals over the last decade may have generated improvements in Japanese technological resource reallocation that resulted in larger aggregate productivity gains than those realized from both within firm technological improvements and domestic reallocation of activity.

6. Conclusion

This paper applied a simulation framework to quantitatively investigate how multinationals respond to declining FDI barriers and the implications for aggregate productivity. Our work makes the following contributions. First, using a detailed dataset of Japanese manufacturing firms, we found that the sales and entry patterns were consistent with a heterogeneous firm model of multinationals facing both fixed and variable barriers. An adapted version of the EKK model was capable of replicating both in-sample and out-of-sample activity, supporting both the internal and external validity of the model. Second, our counterfactual simulations illustrated how falling FDI barriers could lead to important changes in intra-industry reallocations. Consistent with the structural changes observed in the data, more efficient firms disproportionately expand while less efficient firms suffer net contractions. Lastly, we found that the type of FDI barriers has significantly different consequences for aggregate productivity improvements, whereby reductions in variable FDI costs generate more pronounced intra-industry reallocations than reductions in fixed FDI costs.

Our numerical simulations are the first attempt to measure the impact of FDI globalization on intra-industry reallocations and its implications for aggregate productivity. We provide important quantitative evidence that falling FDI barriers have contributed to structural changes in

 $^{^{25}}$ Aggregate productivity estimates vary but fall within the range of 0-2% per year for the last two decades (Inui et al., 2009).

Japan's manufacturing sector. As investment barriers continue to decline across the world, our work suggests that the further restructuring towards multinational production in Japan and other advanced economies will be an important source of global economic development.

Our application of the EKK framework to multinational production is a promising approach to quantify the aggregate implications of FDI integration. Further improvements in this micro-simulation framework could lead to a rich computational tool useful for policy analysis. Since our application focused exclusively on FDI, a natural extension is to combine the exporting and FDI activities into an integrated micro-simulation framework. This would allow the analysis to capture the role of intra-firm trade in multinational production. Another extension is to apply the model to more FDI-specific policy issues. Moving beyond ad-hoc hypothetical reductions in FDI barriers, the model could be used to investigate changes in foreign taxes on multinationals and reductions in FDI specific regulations. Lastly, as our work has focused primarily on the manufacturing sector, an examination of the heavily regulated service sector would also be an important case study of interest for policy makers.

References

- Arita, S. and K. Tanaka. 2011. "Simulating Heterogeneous Multinational Firms," *RIETI Discussion Paper Series 11-E-025*.
- Arkolakis, C. 2010. "Market Penetration Costs and the New Consumers Margin in International Trade." *Journal of Political Economy*, 118(6), 1151-1199.
- Aw, B.Y. and Y. Lee. 2008. "Firm Heterogeneity and Location Choice of Taiwanese Multinationals." *Journal of International Economics*, 75(1), 167-79.
- Bernard, A.B., J. Eaton, J.B. Jensen and S. Kortum. 2003. "Plants and Productivity in International Trade." *American Economic Review*, 93(4), 1268-90.
- Blonigen, B.A. 2001. "In Search of Substitution between Foreign Production and Exports." *Journal of International Economics*, 53(1), 81-104.
- Brainard, S.L. 1997. "An Empirical Assessment of the Proximity-Concentration Trade-Off between Multinational Sales and Trade." *American Economic Review*, 87(4), 520-44.
- Burstein, A. and A. Monge-Naranjo. 2009. "Foreign Know-How, Firm Control, and the Income of Developing Countries." *Quarterly Journal of Economics*, 124(1), 149-95.
- Chen, M.X. and M.O. Moore. 2010. "Location Decision of Heterogeneous Multinational Firms." *Journal of International Economics*, 80(2), 188-99.
- Chor, D. 2010. "Unpacking Sources of Comparative Advantage: A Quantitative Approach." Journal of International Economics, 82(2), 152-67.
- Dekle, R., J. Eaton and S. Kortum. 2008. "Global Rebalancing with Gravity: Measuring the Burden of Adjustment." *IMF Staff Papers*, 55(3), 511-40.
- di Giovanni, J. and A. Levchenko. 2010. "Firm Entry, Trade, and Welfare in Zipf's World," *NBER Working Paper No. 16313.*
- Eaton, J. and S. Kortum. 2002. "Technology, Geography, and Trade." *Econometrica*, 70(5), 1741-79.
- Eaton, J., S. Kortum and F. Kramarz. 2011. "An Anatomy of International Trade: Evidence from French Firms." *Econometrica, forthcoming.*
- Economist. 2009. "Turning their backs on the world." The Economist.
- Foster, L., J.C. Haltiwanger and C.J. Krizan. 2001. "Aggregate Productivity Growth. Lessons from Microeconomic Evidence," In C. R. Hulten, E. R. Dean, and M. J. Harper (Ed.), *New Developments in Productivity Analysis*, University of Chicago Press,
- Garetto, S. 2010. "Input Sourcing and Multinational Production." mimeo.
- Girma, S., R. Kneller and M. Pisu. 2005. "Exports versus FDI: An Empirical Test." *Review of World Economics*, 141(2), 193-218.
- Gormsen, C. 2011. "The Declining Barriers to Foreign Direct Investment and How to See Them," mimeo.
- Grossman, G.M., E. Helpman and A. Szeidl. 2006. "Optimal Integration Strategies for the Multinational Firm." *Journal of International Economics*, 70(1), 216-38.
- Head, K and J. Ries. 2003. "Heterogeneity and the FDI Versus Export Decision of Japanese Manufacturers." *Journal of the Japanese and international economies*, 17(4), 448-67.
- Helpman, E., M.J. Melitz and S.R. Yeaple. 2004. "Export Versus FDI with Heterogeneous Firms." *American Economic Review*, 94(1), 300-16.
- Inui, T., R. Kneller, T. Matsuura and D. McGowan. 2009. "Globalization, Productivity and Plant Exit: Evidence from Japan." *RIETI Discussion Paper Series 09-E-048*.
- Irarrazabal, A., A. Moxnes and L.D. Opromolla. 2010. "The Margins of Multinational Production and the Role of Intra-Firm Trade." mimeo.
- Kang, K. 2008. "How Much Have Been the Export Products Changed from Homogeneous to Differentiated? Evidence from China, Japan, and Korea." *China Economic Review*, 19(2), 128-37.
- Keane, M. and K.I. Wolpin. 2007. "Exploring the Usefulness of a Nonrandom Holdout Sample for Model Validation: Welfare Effects on Female Behavior." *International Economic*

Review 48(4), 1351-78

- Keller, W. and S. R. Yeaple. 2008. "Global Production and Trade in the Knowledge Economy." *NBER Working Paper No. 14626.*
- Markusen, J.R. 2004. *Multinational firms and the theory of international trade*. Cambridge, MA: MIT Press.
- McFadden, D. 1989. "A Method of Simulated Moments for Estimation of Discrete Response Models without Numerical Integration." *Econometrica*, 57(5), 995-1026.
- McGrattan, E. R. and E. C. Prescott. 2009. "Openness, Technology Capital, and Development." *Journal of Economic Theory*, 144(6), 2454-76.
- Melitz, M. 2003. "The Impact of Trade on Aggregate Industry Productivity and Intra-Industry Reallocations." *Econometrica*, 71(6), 1695-725.
- Nelder, J.A. and R. Mead. 1965. "A Simplex Method for Function Minimization." *Computer Journal*, 7(4), 308-313.
- Pavenik, N. 2002. "Trade Liberalization, Exit, and Productivity Improvements: Evidence from Chilean Plants." *Review of Economic Studies*, 69(1), 245-276.
- Petrin, A. T., K. White and J.P. Reiter. 2011. "The Impact of Plant-Level Resource Reallocations and Technical Progress on U.S. Macroeconomic Growth." *Review of Economic Dynamics*, 14(1), 3-26.
- Ramondo, N. 2011. "A Quantitative Approach to Multinational Production." mimeo.
- Ramondo, N. and A. Rodriguez-Clare. 2009. "Trade, Multinational Production, and the Gains from Openness." NBER Working Paper No. 15604.
- Tomiura, E. 2007. "Foreign Outsourcing, Exporting, and FDI: A Productivity Comparison at the Firm Level." *Journal of International Economics*, 72(1), 113-27.
- UNCTAD. 1990/2011. World Investment Report.
- Wakasugi, R., Y. Todo, H. Sato, S. Nishioka, T. Matsuura, B. Ito and A. Tanaka. 2008. "The Internationalization of Japanese Firms: New Findings Based on Firm-Level Data." *RIETI Discussion Paper Series 08-E-036*
- World Bank Group. 2010. "Investing Across Borders", Washington D.C.
- Yamashita, N. and K. Fukao. 2010. "Expansion Abroad and Jobs at Home: Evidence from Japanese Multinational Enterprises." *Japan and the World Economy*, 22(2), 88-97.
- Yeaple, S.R. 2009. "Firm Heterogeneity and the Structure of US Multinational Activity." Journal of International Economics, 78(2), 206-15.

		# All Firı	ns	# Multinationals		
Initial Size	Ye	<u>ear</u>	Change from	<u>Y</u>	ear	Change
(percentile)	1996	2006	1996	1996	2006	from 1996
0-10	1,411	1,376	-35	0	3	3
10-20	1,410	1,276	-134	5	13	8
20-30	1,411	1,178	-233	3	20	17
30-40	1,412	1,229	-183	11	40	29
40-50	1,412	1,202	-210	16	36	20
50-60	1,414	1,191	-223	27	73	46
60-70	1,411	1,299	-112	51	113	62
70-80	1,413	1,229	-184	75	185	110
80-90	1,412	1,409	-3	184	359	175
90-99	1,270	1,309	39	464	677	213
99-100	141	157	16	124	137	13
Total	14,117	12,855	-1,262	960	1,656	696

Table 1. Firm Distribution by Initial Size of Year 1996

Notes: Percentile bins are determined by parent firms' total sales in 1996; all firms include domestic and multinational firms in manufacturing; we exclude the firms whose domestic sales are missing.

Source: Basic Survey of Japanese Business Structure and Activities, and Basic Survey of Overseas Business Activities from METI.

	Domestic Production per Firm			Foreign Pro	Foreign Production per Multinational			Total Production per Firm	
Size Interval	Ye	ear and a second se	Change from	Y	ear_	Change	Y	ear	Change
(percentile)	1996	2006	1996	1996	2006	from 1996	1996	2006	from 1996
0-10	0.86	0.83	-0.03	0.00	0.10	0.10	0.86	0.83	-0.03
10-20	1.47	1.43	-0.04	0.17	0.25	0.08	1.47	1.43	-0.04
20-30	2.01	1.99	-0.02	0.31	0.27	-0.04	2.01	2.00	-0.01
30-40	2.64	2.66	0.02	0.24	0.57	0.33	2.64	2.68	0.03
40-50	3.49	3.56	0.06	0.70	0.81	0.11	3.50	3.58	0.08
50-60	4.67	4.87	0.19	0.65	1.04	0.39	4.68	4.93	0.25
60-70	6.54	7.02	0.48	1.09	1.26	0.18	6.58	7.15	0.57
70-80	10.1	11.0	0.97	1.47	2.17	0.69	10.2	11.4	1.24
80-90	19.1	21.4	2.34	2.93	4.23	1.30	19.5	22.6	3.13
90-99	86.9	100.1	13.2	19.2	29.7	10.6	93.9	116.7	22.7
99-100	1,664	1,570	-94.3	306.8	638.8	331.9	1,934	2,143	208.9
All	29.5	30.1	0.59	49.7	57.3	7.65	32.9	37.5	4.59

 Table 2 Average Volume of Production by Size Interval in 1996 and 2006

Notes: Percentile bins are determined by parent firms' total sales in each year, separately; production is in billions of 2006 Yen; domestic production includes domestic and export sales; total production includes both domestic and foreign production.

Sources: Basic Survey of Japanese Business Structure and Activities, and Basic Survey of Overseas Business Activities from METI.

	(1)	(2)	(3)	(4)
Markets	Markets with over 10 affiliates	All Markets	Markets with over 10 affiliates	Markets with over 10 affiliates
Year	2006	2006	2006	1996
Moments	All	All	No Pecking Order String	All
Variable				
$ ilde{ heta}$	1.99	2.12	1.95	2.13
(size dispersion)	(0.43)	(0.95)	(0.64)	(0.53)
σ_a	1.64	1.64	1.66	1.36
(variance of sales shock)	(0.07)	(0.10)	(0.08)	(0.11)
σ_η	0.39	0.52	0.34	0.45
(variance of entry shock)	(0.31)	(0.16)	(0.42)	(0.43)
ρ	-0.62	-0.55	-0.64	-0.99
(correlation of sales and entry shocks)	(0.34)	(0.25)	(0.51)	(0.56)

Table 3. Estimation Results of Structural Parameters

Notes: Figures indicate parameter estimates of each variable; parentheses are bootstrapped standard errors from the initial fixed parameter estimates with 1000 repetitions; each bootstrapping simulates 100,000 firms and uses randomly sampled Japanese firms.

		(1)	(2)	(3)
Country	Code	Relative Change in Real Wage	% Change in Japanese Firm Entry	% Change in Japanese Firm Sales
Argentina	ARG	1.08	93.1	81.1
Australia	AUS	1.14	84.8	98.8
Belgium	BEL	1.46	39.5	155.9
Brazil	BRA	1.05	105.3	88.8
Canada	CAN	1.17	73.9	116.6
China	CHN	1.25	64.6	71.5
Czech Republic	CZE	1.23	67.4	79.0
France	FRA	1.18	66.9	125.4
Germany	DEU	1.09	98.2	101.4
Hong Kong	HKG	4.92	-42.1	277.2
Hungary	HUN	1.31	63.3	109.1
India	IND	1.02	111.5	82.8
Indonesia	IDN	1.03	98.4	72.1
Italy	ITA	1.07	100.8	93.4
Japan	JPN	1.01	-2.67	-3.45
South Korea	KOR	1.05	109.4	89.2
Malaysia	MYS	1.22	73.3	78.3
Mexico	MEX	1.06	97.3	80.2
Netherlands	NLD	1.31	18.6	146.2
New Zealand	NZL	1.21	70.7	96.7
Philippines	PHL	1.11	88.3	89.0
Poland	POL	1.11	87.4	87.4
Singapore	SGP	1.74	16.4	192.1
South Africa	ZAF	1.07	96.6	84.0
Spain	ESP	1.12	91.0	100.5
Sweden	SWE	1.31	59.3	128.2
Taiwan	TWN	1.10	94.8	118.4
Thailand	THA	1.24	64.9	71.9
Turkey	TUR	1.06	99.3	84.9
United Kingdom	GBR	1.23	56.9	139.3
United States	USA	1.07	98.4	99.7
Vietnam	VNM	1.17	73.2	72.8
Average		1.29	72.5	103.4

 Table 4.1 Aggregate Outcomes of Counterfactual 20% Reductions in FDI Barriers

Table 4.2 Counternactual Results for Aggregate Firm Activities							
Variable	Baseline	Counterfactual Change from Baseline	% Change from Baseline				
Number of Firms:							
All	13,123	-350	-2.67				
Multinationals	1,511	1,004	66.4				
Aggregate Production:							
Domestic	394.3	-13.6	-3.45				
Foreign	99.8	107.1	107.3				
Total	494.1	93.5	18.9				

Table 4.2 Counterfactual Results for Aggregate Firm Activities

Note: Production is in trillions of Yen.

	# Multinationals		# All Firms		
Initial Productivity Group (percentile)	Baseline	Counterfactual Change from Baseline	Baseline	Counterfactual Change from Baseline	
0-10	0	6	1,313	-345	
10-20	0	21	1,312	-2	
20-30	0	34	1,313	-1	
30-40	2	53	1,312	-1	
40-50	22	62	1,312	-1	
50-60	45	82	1,312	0	
60-70	79	118	1,312	0	
70-80	146	177	1,313	0	
80-90	322	259	1,312	0	
90-99	766	191	1,181	0	
99-100	130	0	130	0	
Total	1,511	1,004	13,123	-350	

Table 4.3 Counterfactual Results for the Extensive Margin

	Domestic Pro	duction per Firm	Foreign Production	on per Multinational	Total Production per Firm	
Initial Productivity Group (percentile)	Baseline	Counterfactual Change from Baseline	Baseline	Counterfactual Change from Baseline	Baseline	Counterfactual Change from Baseline
0-10	13.7	-0.14	0.00	1.34	13.7	-0.13
10-20	14.6	-0.37	0.00	1.27	14.6	-0.35
20-30	16.0	-0.42	0.00	1.54	16.0	-0.38
30-40	17.9	-0.52	0.69	0.90	17.9	-0.45
40-50	19.5	-0.48	0.75	1.31	19.5	-0.36
50-60	22.0	-0.50	1.04	1.50	22.0	-0.29
60-70	26.1	-0.67	1.58	1.92	26.2	-0.24
70-80	31.5	-0.67	2.36	3.14	31.7	0.42
80-90	41.4	-0.88	5.01	6.16	42.6	2.84
90-99	76.5	-1.61	32.0	31.44	97.3	29.1
99-100	293.0	-6.17	563.2	486.7	855.4	481.3
All	30.0	-0.24	66.1	16.22	37.7	8.4

Note: Production is measured in billions of Yen.

Variable	Decelino -	Counterfactual % Change from Baseline		
variable	Basennie –	Variable Cost	Fixed Cost	
Number of Firms:				
All	13,123	-1.32	-0.37	
Multinationals	1,511	28.0	30.56	
Aggregate Production:				
Domestic	394.3	-0.74	-0.28	
Foreign	99.8	64.0	19.7	
Total	494.1	12.11	4.41	

Table 5.1 Counterfactual Aggregate Results for Variable and Fixed Costs

Note: Production is in trillions of Yen.

	# Multinationals		# All I	Firms
Initial Productivity Group (percentile)	Variable Cost	Fixed Cost	Variable Cost	Fixed Cost
0-10	3	3	-171	-48
10-20	9	11	0	0
20-30	11	14	0	0
30-40	14	15	0	0
40-50	20	22	0	0
50-60	29	31	0	0
60-70	45	50	0	0
70-80	70	79	0	0
80-90	114	123	0	0
90-99	100	108	0	0
99-100	0	0	0	0
Total	415	455	-172	-48

Table 5.2 Counterfactual Changes from Baseline at the Extensive Margin

Note: Baseline is fixed as in Table 4.3.

	Domestic Production per Firm		Foreign Production per Multinational		Total Production per Firm	
Initial Productivity Group (percentile)	Variable Cost	Fixed Cost	Variable Cost	Fixed Cost	Variable Cost	Fixed Cost
0-10	0.08	0.10	0.86	0.83	0.08	0.10
10-20	-0.03	0.06	0.95	0.95	-0.03	0.07
20-30	-0.07	0.06	0.61	0.27	-0.05	0.07
30-40	-0.04	0.07	0.60	0.21	-0.02	0.09
40-50	-0.04	0.07	0.69	0.23	0.00	0.10
50-60	-0.05	0.10	0.79	0.28	0.03	0.15
60-70	-0.03	0.12	1.13	0.38	0.13	0.22
70-80	-0.03	0.14	1.60	0.60	0.35	0.40
80-90	-0.05	0.18	3.50	1.21	1.53	1.06
90-99	-0.08	0.34	18.21	6.03	15.76	7.65
99-100	-0.35	1.52	314.8	82.5	313.5	83.8
All	0.18	0.20	18.4	1.65	5.11	1.84

Table 5.3 Co	ounterfactual	Changes from	Baseline at th	e Intensive	Margin

Note: Production is measured in billions of Yen.

Table 6. Summary of Counterfactual Results

			Counterfactual Reductions in Variable Barriers:			
	Actual %	25%	20%	15%	20%	0%
	Change between		Counterfactua	ll Reductions in F	ixed Barriers:	
Counterfactual Percentage Change	1996 and 2006	25%	20%	15%	0%	20%
Volume of Production						
Domestic	-7.0	-4.6	-3.4	-4.1	-0.7	0.3
Foreign	99.0	148.1	106.2	71.0	64.0	19.7
Total	4.0	25.5	16.8	11.2	12.1	4.4
Production Growth by Bottom 80% Firms*	6.5	-4.4	18.9	-4.6	-0.9	0.4
Production Growth by Top 20% Firms*	14.7	36.3	30.0	21.2	21.0	7.8
Number of Firms						
All	-9.0	-3.5	-2.7	-2.4	-1.3	-0.4
Multinational	72.0	85.9	66.4	48.6	28.0	30.6
Aggregate Productivity						
Contributed by:		37.4	27.2	20.1	18.1	4.1
Market-Share Reallocation	n.a.	35.9	26.0	19.1	17.6	4.0
Exit of Least Productive Firms	n.a.	1.5	1.2	1.0	0.5	0.1

Notes: Figures are in percentage; aggregate productivity is a percentage change in industry-level productivity of simulated firms resulting from market-share reallocation and exit effects; * refers to a percentage point change.



















APPENDIX

A. The Price Index

Following Eaton et al. (2011), a representative consumer in market *n* faces the following price index, P_n :

$$P_n = \overline{m} \left[\iint \left(\sum_{i=1}^N \int_0^{\overline{c}_{ni}(\eta)} \alpha_n(j) c^{-(\sigma-1)} d\mu_{ni}(c) \right) g(\alpha, \eta) d\alpha d\eta \right]^{-1/(\sigma-1)}$$
(A1)

where $g(\alpha, \eta)$ is a joint density function from which we draw demand and entry shocks specific to individual producers, $\alpha_n(j)$ and $\eta_n(j)$. To solve for the price index, we first rewrite the integral over $\mu_{ni}(c) = T_i(w_n d_{ni})^{-\theta}(c)^{\theta}$ in equation (A1):

$$P_n = \overline{m} \left[\iint \left(\alpha_n(j) \sum_{i=1}^N \Phi_{ni} \int_0^{\overline{c}_{ni}(\eta)} \theta c^{\theta - \sigma} dc \right) g(\alpha, \eta) d\alpha d\eta \right]^{-1/(\sigma - 1)}$$
(A2)

where $\Phi_{ni} = T_i (w_n d_{ni})^{-\theta}$. Note that $d\mu_{ni}(c) = \Phi_{ni}\theta c^{\theta-1}dc$. From the laws of integration, we can solve for the integral over *c*:

$$P_n = \bar{m} \left[\iint \left(\alpha_n(j) \sum_{i=1}^N \Phi_{ni} \frac{\theta}{\theta - (\sigma - 1)} \bar{c}_{ni}(\eta)^{\theta - (\sigma - 1)} \right) g(\alpha, \eta) d\alpha d\eta \right]^{-1/(\sigma - 1)}$$
(A3)

Substituting the threshold unit cost, $\bar{c}_{ni}(\eta)$, into equation (A3), and rearranging gives:

$$P_{n} = \bar{m} \left[\kappa_{0} \iint \left(\alpha_{n}(j) \sum_{i=1}^{N} \Phi_{ni} \left(\left(\eta_{n}(j) \frac{X_{n}}{\sigma E_{ni}} \right)^{1/(\sigma-1)} \frac{P_{n}}{\bar{m}} \right)^{\theta-(\sigma-1)} \right) g(\alpha, \eta) d\alpha d\eta \right]^{-1/(\sigma-1)}$$
(A4)

where $\kappa_0 = \frac{\theta}{\theta - (\sigma - 1)}$.

By moving the price-index term to the left side, we obtain the following:

$$P_n = \overline{m}(\kappa_1 \psi_n)^{\frac{-1}{\theta}} X_n^{\left(\frac{1}{\theta} - \frac{1}{\sigma - 1}\right)}$$
(A5)

where:

$$\kappa_{1} = \kappa_{0} \iint \alpha_{n}(j)\eta_{n}(j)^{\frac{\theta - (\sigma - 1)}{\sigma - 1}}g(\alpha, \eta)d\alpha d\eta$$

$$\psi_{n} = \sum_{i=1}^{N} \Phi_{ni} (\sigma E_{ni})^{\frac{-[\theta - (\sigma - 1)]}{\sigma - 1}}.$$

B. Respecification for Simulation

Substituting the price index in equation (A5) into equation (9), and rearranging gives:

$$\bar{c}_{ni}(j) = \left(\frac{\eta_n(j)}{\sigma E_{ni}}\right)^{1/(\sigma-1)} \left(\frac{x_n}{\kappa_1 \psi_n}\right)^{\frac{1}{\theta}}$$
(B1)

Replacing equation (B1) into equation (10), we arrive at:

$$X_{ni}(\alpha, \eta) = \alpha_n(j)\eta_n(j)^{\frac{\theta - (\sigma - 1)}{\sigma - 1}} \frac{\kappa_0}{\kappa_1} \pi_{ni} X_n$$
(B2)

where:

$$\pi_{ni} = \frac{\Phi_{ni}(\sigma E_{ni})^{\frac{-[\theta - (\sigma - 1)]}{\sigma - 1}}}{\psi_n}$$

From this point, we focus on a country pair between a host country *i* and a home country J, where J denotes Japan. We proceed to express equations (B1) and (B2) in terms of a standardized unit cost. Rearranging equation (11) gives $z_J(j) = T_J^{\frac{1}{\theta}}u(j)^{\frac{-1}{\theta}}$. Substituting this expression into equation (1), we obtain $c_{nJ}(j) = \left(\frac{u(j)}{\Phi_{nJ}}\right)^{\frac{1}{\theta}}$. The threshold unit cost for entry is then given by:

$$\bar{c}_{nJ}(j) = \left(\frac{\bar{u}(j)}{\Phi_{nJ}}\right)^{\frac{1}{\theta}}$$
(B3)

Using equations (B1) and (B3), we can rewrite the entry condition as follows:

$$u(j) \le \bar{u}(j) \tag{B4}$$

where:

$$\overline{u}(j) = \eta_n(j)^{\frac{\theta}{\sigma-1}} \frac{X_n \pi_{nJ}}{\kappa_1 \sigma E_{nJ}}$$
(B5)

We can also express the sales condition in terms of the standardized unit cost. Replacing the unit cost for firm j and the threshold unit cost in equation (10), we obtain:

$$X_{nJ}(j) = \frac{\alpha_n(j)}{\eta_n(j)} \sigma E_{nJ} \left(\frac{\overline{u}(j)}{u(j)}\right)^{\frac{\sigma-1}{\theta}}$$
(B6)

Finally, we parameterize the entry and sales conditions by connecting the conditions with actual data on aggregate multinational production. We first express the total number of firm entries to market *n* from Japan as:

$$N_{nJ} = \int \left[\mu_{nJ}(\bar{c}_{nJ}(\eta)) \right] g_2(\eta) d\eta \tag{B7}$$

Substituting $\mu_{nJ}(c) = \Phi_{nJ}(\bar{c}_{nJ}(j))^{\theta}$ and equation (B1) into equation (B7), we obtain: $N_{nJ} = \kappa_0 \frac{x_n \pi_{nJ}}{c_{nJ}}$

$$N_{nJ} = \kappa_2 \frac{\kappa_n \kappa_{nJ}}{\kappa_1 \sigma E_{nJ}} \tag{B8}$$

Replacing equation (B8) into (B5) gives:

$$\bar{u}(j) = \eta_n(j)^{\frac{\theta}{\sigma-1}} \frac{N_{nJ}}{\kappa_2}$$
(B9)

This expression corresponds to equation (12).

We then proceed to connect the sales condition with data by replacing $X_n \pi_{nJ}$ in equation (B8) with data. Integrating equation (B2) across the joint density $g(\alpha, \eta)$, total sales by firms from Japan in country *n* are given by:

$$X_{nJ} = \iint \alpha_n(j)\eta_n(j)^{\frac{\theta - (\sigma - 1)}{\sigma - 1}} \frac{\kappa_0}{\kappa_1} \pi_{ni} X_n g(\alpha, \eta) d\alpha d\eta$$
(B10)

Rearranging gives $X_{nJ} = \pi_{ni}X_n$. Substituting this expression in equation (B8), we obtain:

$$\sigma E_{nJ} = \frac{\kappa_2}{\kappa_1} \frac{\kappa_{nJ}}{N_{nJ}} \tag{B11}$$

Replacing equation (B11) into equation (B6) gives:

$$X_{nJ}(j) = \frac{\alpha_n(j)}{\eta_n(j)} \overline{X}_{nJ} \frac{\kappa_2}{\kappa_1} \left(\frac{\overline{u}(j)}{u(j)}\right)^{\frac{\sigma-1}{\theta}}$$
(B12)

where $\overline{X}_{nJ} = X_{nJ}/N_{nJ}$. This expression corresponds to equation (13).

C. Estimation Framework

Our estimation procedure is based on Eaton et al. (2011). We proceed to (1) simulate artificial firms under a particular set of parameter values, (2) construct a set of four moments, and (3) search an optimal set of structural parameters by the simulated method of moments. Throughout this section, we denote home country *i*, by *J* (Japan) and an artificial firm *s* by s = 1, 2, ..., S.

C.1 Simulation of Multinational Firms

We use the entry and sales conditions in equations (12) and (13) to simulate individual multinational activities. An artificial producer s is generated from its efficiency draw, u(s), sales shock, $\alpha_n(s)$, and entry shock $\eta_n(s)$. These simulated draws are generated from the stochastic distributions as dictated by the four structural parameters in Θ : heterogeneity in observed sales $\tilde{\theta}$, variances in sales σ_a and entry shocks σ_η , and correlation between these shocks ρ . With aggregate data on the number of Japanese firms investing in country n, N_{nJ} , and their aggregate sales X_{nJ} in that country and Θ , we can produce an artificial dataset of heterogeneous multinationals on the location and sales of their foreign affiliates.

To simulate the total number S of firms, we first fix Θ and construct realizations for standardized unit cost, u(s), for each firm s. Demand and entry shocks, $\alpha_n(s)$ and $\eta_n(s)$, are

generated from a joint lognormal distribution for each firm *s* across country n.²⁶ These realizations of stochastic components are fixed throughout the estimation. We then compute the entry hurdle condition (12) for each artificial firm across each market and define an indicator variable *Z* of whether each firm establishes local production in each market *n*:

$$Z_{nJ}(s) = \begin{cases} 1 & \text{if } u(s) \le \bar{u}_n(s) \\ 0 & \text{otherwise} \end{cases}$$
(C1)

When the entry indicator is equal to one, a simulated firm from country J (Japan) enters the market. Conditional on entry, we compute its sales in market n according to the sales condition (13). We then construct the matrix of sales for each artificial firm across each market that indicates where each artificial firm sets up a foreign affiliate and how much sales it generates in that country.

C.2 Moments

With an artificial dataset of individual multinational entry and sales, we construct a vector of moments. Each moment is defined as the share of multinational parent firms that fall into a set of mutually exclusive bins. We denote N^k as the number of actual firms achieving an outcome k in the actual data, and \hat{N}^k as the number of simulated firms achieving the same outcome. For each moment, the number of simulated firms falling into each outcome is weighted as follows:

$$\widehat{N}^{k} = \frac{1}{S} \sum_{s=1}^{S} \overline{u}(s) Z_{nJ}^{k}(s)$$
(C2)

where $\bar{u}(s)$ is the importance weight of each simulated firm. We define a vector of deviations between actual and artificial moments for outcome k:

$$y(\Theta) = m^k - \hat{m}^k(\Theta) \tag{C3}$$

Following the theoretical implications of the model, we choose four moment conditions: pecking order strings, affiliate sales distributions across markets, parent sales distribution in Japan, and multinational production intensity. Corresponding to these moments, we also construct a vector of moments from the actual data on Japanese multinational firms. As described in empirical regularities of Japanese firms, we seek to capture these characteristics for simulated multinationals. Given that N is the number of foreign markets penetrated by Japanese firms, each moment is defined as follows:

1. Pecking order string

We compute the share of multinationals entering each possible combination of the five most popular countries for Japanese multinationals: China, the United States, Thailand, Taiwan, and Indonesia in 2006. Determining the market popularity by the number of their foreign affiliates

$$\begin{bmatrix} \ln \alpha_n(s) \\ \ln \eta_n(s) \end{bmatrix} \sim \mathbb{N} \begin{bmatrix} \begin{pmatrix} 0 \\ 0 \end{pmatrix}, & \begin{pmatrix} \sigma_\alpha^2 & \rho \sigma_\alpha \sigma_\eta \\ \rho \sigma_\alpha \sigma_\eta & \sigma_\eta^2 \end{pmatrix} \end{bmatrix}.$$

To draw these realizations of random shocks, we use the Choleski decomposition factor to construct $\ln a$ and $\ln \eta$ with:

$$\begin{bmatrix} \ln \alpha_n(s) \\ \ln \eta_n(s) \end{bmatrix} \sim \begin{bmatrix} \left(\sigma_a \sqrt{1 - \rho^2} & \rho \sigma_a \\ 0 & \sigma_h \end{bmatrix} \begin{bmatrix} \left(a_n(s) \\ h_n(s) \end{bmatrix} \end{bmatrix}.$$

To avoid drawing the firms that end up not selling anywhere, productivity draws are bounded to the firms that sell in Japan and at least one foreign market. In doing so, we use the importance sampling from $u(s) = v(s)\bar{u}_J^*(s)$, where random realizations of v(s) are independently drawn from a uniform distribution over interval [0,1] and $\bar{u}_J^*(s)$ is the firm-specific hurdle for entering the Japanese market and at least one foreign market. As this measure serves as a sampling weight, we ensure that all draws are to be $u(s) \leq \bar{u}_J^*(s)$, which corrects an upward bias in generating more efficient firms.

²⁶ We assume that $\alpha_n(s)$ and $\eta_n(s)$ have a joint bivariate lognormal distribution:

we set each string such that multinationals entering the most popular market (China) invest in less popular markets progressively. Then, we make another combination for that string such that multinationals entering the first (China) and third (Thailand) invest in less popular markets progressively. By adding up all possible combinations, we have 2^5 moments.

2. Sales distributions across foreign markets

We calculate q^{th} percentiles for multinational sales in each market *n*, for q = 50, 75, 100. For each set of firms that enter market *n*, we use these percentiles to set up sales intervals. We then calculate the share of multinationals that fall into each of these bins. The q^{th} percentiles are calculated from the actual data and the simulated firms are set according to these bins. ($N \times 3$ moments).

3. Distribution of multinational parent sales in Japan

This moment links the level of sales in Japan to the set of firms that enter market *n*. We calculate q^{th} percentiles (q = 50, 75, 100) over domestic sales in Japan for each set of firms that enter market *n*. These intervals are calculated from the actual data. We then assign the firms that fall into these bins and calculate the share of multinationals ($N \times 3$ moments).

4. Multinational production intensity

We make two intervals for firms whose ratio of sales in market n to sales in Japan is below and above the 50th percentile. Then, we compute the share of simulated firms that sell in each market n and fall into either of these percentiles.

C.3 Simulated Method of Moments

Equation (C3) shows that an optimal set of parameters can be judged from the distance between actual and artificial moments, with the smaller distance indicating better parameters. To estimate the parameters, we employ simulated method of moments as introduced by McFadden (1989). This estimation method matches moments of simulated and real data, and searches for a set of parameters that minimizes the total distance between them. Under the true set of parameter values, Θ_0 , the following moment condition is assumed to hold:

$$\mathbf{E}[y(\Theta_0)] = 0 \tag{C4}$$

An objective function is specified under the following weighted quadratic form:

$$\widehat{\Theta} = \arg\min_{\Theta} \{ y(\Theta)' W y(\Theta) \}$$
(C5)

where $\widehat{\Theta}$ is a set of estimated parameters, and W is assumed to be an identity matrix in practice.

To search for the parameters that best fit the model, we employ the Nelder-Mead simplex method (Nelder and Mead, 1965). To mitigate optimization errors, we introduce random variations to the starting values and repeat the minimization algorithm for a fixed set of artificial and real moments 1000 times. Finally, we take the optimal parameters that give the minimum distance.

The search procedure above should provide the best fitting parameters. However, they may be subject to sampling errors of real Japanese multinationals and simulation errors of artificial multinationals. To address these issues, we compute bootstrapped standard errors for the initial optimal estimates. First, we resample Japanese multinational firms from the actual dataset with replacement. A dataset of simulated multinationals is also generated from a new set of idiosyncratic parameters $u^{b}(s)$, $\alpha_{n}^{b}(s)$, and $\eta_{n}^{b}(s)$. Then, we follow the simulated method of moments to estimate a new set of structural parameters, $\widehat{\Theta}_{b}$. Repeating 25 times, we calculate:

$$V(\Theta) = \frac{1}{25} \sum_{1}^{25} (\widehat{\Theta}_{b} - \widehat{\Theta}^{*}) (\widehat{\Theta}_{b} - \widehat{\Theta}^{*})'$$
(C6)

where $\widehat{\Theta}^*$ is the initial set of the best fitting structural parameters. Taking the square root of the diagonal elements, we obtain the bootstrapped standard errors. They serve to gauge the potential

influence of sampling and simulation errors in the best fitting parameters.

D. Counterfactual Simulations

D.1 General Equilibrium Framework

Following the approach in EKK (2011), each country is endowed with an amount of labor, L_i , which is mobile across sectors but not countries, and has a wage rate W_i . Intermediates are a Cobb-Douglass combination of labor and intermediates with an input bundle of $W_n^{\beta} P_n^{1-\beta}$. Final output is also non-traded and is a Cobb-Douglas combination of manufactures and labor, with manufactures having a share γ .

Profits are earned by the country where the firm is headquartered and consumers own equal shares of each firm headquartered in their country. This allows profits to be redistributed equally among the consumers of their country. Country *i's* total GDP, Y_i^A is equal to its total wages generated from production in the country, $W_i L_i$ and its total profits abroad Π_i . Lastly, countries have FDI deficits D_i , where some countries are net receivers of FDI and some are net providers. Net profits earned in destination n are gross profits X_n/σ minus total entry costs incurred there, where total profits are $\Pi_n^D = \frac{\sigma - 1}{\sigma \theta} X_n$. In equilibrium, firms or multinationals from country *i* earn a share π_{ni} from the total level of profits Π_n^D from each market:

$$\Pi_i = \sum_{n=1}^N \pi_{ni} \Pi_n^D = \frac{\sigma - 1}{\sigma \theta} Y_i \tag{D1}$$

Country i's total GDP, Y_i^A is equal to its total wages generated from production in the country, W_iL_i and its total profits from abroad Π_i :

$$V_i^A = W_i L_i + \Pi_i \tag{D2}$$

Total demand for manufacturing production in country i, X_i , must be equal to final demand and use of manufactures in intermediates:

$$X_i = \gamma \left(Y_i^A \right) + \left[\frac{(1-\beta)(\sigma-1)}{\sigma} \right] X_i \tag{D3}$$

The first component is the share of manufacturing production from final demand and the second component is the share from its use as intermediates. Using the profit expression (D1) and the income equation (D2), the demand equation (D3) can be respecified as:

$$X_{i} = \gamma \left(W_{i}L_{i} + \frac{\sigma - 1}{\sigma \theta} Y_{i} \right) + \left[\frac{(1 - \beta)(\sigma - 1)}{\sigma} \right] X_{i}$$
(D3)

Assuming that gross manufacturing production in market X_n is equal to gross manufacturing of all production with n's technology Y_i plus an FDI sales deficit D_n , we have:

$$Y_i = \frac{[\gamma\sigma(W_iL_i) + [-\beta\sigma + \beta - 1]D_i]\theta}{\sigma\theta - \sigma\gamma + \gamma - (\sigma - 1)(1 - \beta)\theta}$$
(D4)

$$X_n = \frac{[\gamma\sigma(W_iL_i) + [-\beta\sigma + \beta - 1]D_i]\theta}{\sigma\theta - \sigma\gamma + \gamma - (\sigma - 1)(1 - \beta)\theta} + D_n$$
(D5)

With $\pi_{ni} = \frac{\Phi_{ni}(\sigma E_{ni})^{\frac{-[\theta - (\sigma - 1)]}{\sigma - 1}}}{\psi_n}$ from (B2) and $\psi_n = \sum_{i=1}^N \Phi_{ni} (\sigma E_{ni})^{\frac{-[\theta - (\sigma - 1)]}{\sigma - 1}}$ from (A5), we

can write:

$$\pi_{ni} = \frac{T_i \left(W_n^{\beta} P_n^{1-\beta} d_{ni} \right)^{-\theta} (\sigma E_{ni})^{-[\theta - (\sigma - 1)]/(\sigma - 1)}}{\sum_{i=1}^N T_i (W_n^{\beta} P_n^{1-\beta} d_{ni})^{-\theta} (\sigma E_{ni})^{-[\theta - (\sigma - 1)]/(\sigma - 1)}}$$
(D6)

and the following price equation:

$$P_{n} = \bar{m}\kappa_{1}^{-1/\theta} \left[\sum_{i=1}^{N} T_{i} (W_{n}^{\beta} P_{n}^{1-\beta} d_{ni})^{-\theta} (\sigma E_{ni})^{-[\theta - (\sigma - 1)]/(\sigma - 1)} \right]^{-1/\theta} \left(\frac{X_{n}}{W_{n}} \right)^{\left(\frac{1}{\theta}\right) - 1/(\sigma - 1)}$$
(D7)

Equilibrium in the world market for manufacturers, $Y_i = \sum_{n=1}^N \pi_{ni} X_n$, sets up a system of equations where changes in real wages and price levels, W_n and P_n , can be solved following an exogenous decrease in d_{ni} and/or E_{ni} . As is the case for welfare gains from trade in the heterogeneous firm model, the reductions in FDI barriers would lead to welfare gains from a decline in price levels P_n (through better access to cheaper goods) and a rise in real wages W_n

(through increased production from rising productivity).

Following Dekle, Eaton, and Kortum (2008), rather than estimating all the parameters embedded in the model, the equations can be respecified in terms of the counterfactuals and their rate of change. The merit of this method is that it requires no information on the initial level of technology, variable FDI barriers, and fixed costs to entry. Under this specification, the change of wages \hat{W}_n and the changes in manufacturing price indices \hat{P}_n around the world can be solved from exogenous changes in \hat{d}_{ni} or \hat{E}_{ni} (where the hat indicates the ratio of counterfactual to baseline). Given Θ , σ , β , γ and global FDI shares (not just for Japan), a simple iterative algorithm may be used to jointly solve for changes in wages and prices resulting from an exogenous decrease in FDI barriers.

D.2 Data for Counterfactuals

In addition to the benchmark parameters estimated in Θ , the counterfactual exercise requires additional parameters that need to be calibrated. We use Kang's (2008) estimate of the elasticity of substitution for the Japanese manufacturing sector and set $\sigma = 2.1$. For labor shares β , we take the averages reported in UNIDO across countries and set $\beta = 2.4$.

To construct a dataset on bilateral FDI activity, we rely on several data sources. We use the RIETI FDI Database for aggregate sales of foreign affiliates of Japanese multinationals in 2006. For other bilateral pairs, affiliate sales are estimated from the UNCTAD data on FDI stocks and flows for the period 1990-2006. In doing so, we first construct bilateral FDI stocks in 2006 for each country pair. Missing figures for bilateral FDI stocks are approximated by the cumulative stocks of FDI flows over 1990-2006. As FDI flows were negative for certain country pairs and periods, an estimate of FDI stocks in 2006 becomes negative for some country pairs. In this case, we replace zero FDI stocks. When the figures remain missing, we assume zero FDI stocks for the corresponding country pairs.

Furthermore, we estimate total FDI stocks in manufacturing sectors by multiplying the figures by 21%. This figure is an average share of manufacturing FDI as reported in the World Investment Report (2010). Finally, we multiply the FDI stocks by 2.0198 to convert into sales by foreign affiliates. This factor is based on the estimated relationship between FDI stocks and affiliate sales, as shown in the World Investment Report (2010).

We use the World Bank's World Development Indicators for data on GDP and manufacturing value-added. To construct a measure of the production that employs domestic technology, we approximate domestic production from value-added. Following an estimate for the average ratio of total sales to value-added in UNIDO, valued-added is converted into total sales by a multiplying factor of 2.745. Second, we subtract total sales of inward FDI from total domestic sales to discount the contribution of foreign technology. Lastly, FDI deficits are calculated by subtracting total sales of outward FDI from total sales of inward FDI.

No.	Author(s)	Title	
296	Shawn ARITA and Kiyoyasu TANAKA	Simulating Heterogeneous Multinational Firms	2011
295	Abu S. SHONCHOY and Seiro ITO	Ramadan School Holidays as a Natural Experiment:Impacts of Seasonality on School Dropout in Bangladesh	2011
294	Abu S. SHONCHOY	Seasonal Migration and Micro-credit in the Lean Period: Evidence from Northwest Bangladesh	2011
293	Futaba ISHIZUKA	Economic Restructuring and Regional Distribution of Enterprises in Vietnam	2011
292	Miki HAMADA	Market Discipline by Depositors: Impact of Deposit Insurance on the Indonesian Banking Sector	2011
291	MURAKAMI Kaoru	Negotiating Social Assistance: The Case of the Urban Poor in Turkey	2011
290	Kazuhiko OYAMADA and Yoko UCHIDA	Domestic, Vertical, and Horizontal Multinationals: A General Equilibrium Approach using the "Knowledge Capital Model"	2011
289	Miwa YAMADA	Is the Anti-Trafficking Framework Really for the 'Victims'?	2011
288	Yasushi HAZAMA	Determinants of Political Tolerance: A Literature Review	2011
287	Hisao YOSHINO	Strategic Trade Policy and Non-Linear Subsidy -In The Case of Price Competition-	2011
286	Natsuko OKA	Neither Exit nor Voice: Loyalty as a Survival Strategy for the Uzbeks in Kazakhstan	2011
285	Ikuo KUROWA, Kaoru NABESHIMA, and Kiyoyasu TANAKA	Innovation Networks among China, Japan, and Korea: Evidence from Japanese Patent Data	2011
284	Hitoshi SUZUKI	Preliminary Discussions on the Urbanization of Rural Areas in Modern Iran	2011
283	Kozo KUNIMUNE	A Model of Economic Growth with Saturating Demand	2011
282	Etsuyo MICHIDA, Cemal Atici, and Michikazu KOJIMA	Does Quality Matter in the Iron and Scrap Trade?	2011
281	Kazunobu HAYAKAWA, Fukunari KIMURA, and Hyun-Hoon LEE	How Does Country Risk Matter for Foreign Direct Investment?	2011
280	Kazunobu HAYAKAWA and Nobuaki YAMASHITA	The Role of Preferential Trade Agreements (PTAs) in Facilitating Global Production Networks	2011
279	Noriyuki YANAGAWA and Mariko	Ex ante barganing and ex post enforcement in trade credit supply:	2011
278	Yoko ASUYAMA	Skill Sorting, Inter-Industry Skill Wage Premium, and Production Chains: Evidence from India 1999-2000	2011
277	Yoko ASUYAMA	Skill Distribution and Comparative Advantage: A Comparison of	2011
276	Bo MENG, Norihiko YAMANO and Colin WEBB	Application of Factor Decomposition Techniques to Vertical Specialisation Measurement	2011
275	Kazunobu HAYAKAWA	Measuring Fixed Costs for Firms' Use of a Free Trade Agreement: Threshold Regression Approach	2011
274	Kenmei TSUBOTA, Yujiro KAWASAKI	Myopic or farsighted: Bilateral Trade Agreements among three symmetric countries	2011
273	Ayako OBASHI, Kazunobu HAYAKAWA, Toshiyuki MATSUURA, Kazuyuki MOTOHASHI	A Two-dimensional Analysis of the Impact of Outward FDI on Performance at Home: Evidence from Japanese Manufacturing Firms	2010
272	Kazunobu HAYAKAWA, Hyun- Hoon LEE, Donghyun PARK	Investment Promotion Agencies: Do They Work?	2010
271	Takeshi INOUE, Shigeyuki HAMORI	An Empirical Analysis on the Efficiency of the Microfinance Investment Market	2010

~ Previous IDE Discussion Papers ~

No.	Author(s)	Title	
270	Bo MENG, Norihiko YAMANO, and Colin Webb	Vertical Specialisation Indicator Based on Supply-Driven Input- Output Model	2010
269	Quoc Hung Nguyen	International Real Business Cycles: A Re-Visit	2010
268	Yoko ASUYAMA, Dalin CHHUN, Takahiro FUKUNISHI, Seiha NEOU, and Tatsufumi YAMAGATA	Firm Dynamics in the Cambodian Garment Industry: Firm Turnover, Productivity Growth, and Wage Profile under Trade Liberalization	2010
267	Kazunobu HAYAKAWA, Hyun- Hoon LEE, Donghyun PARK	The Role of Home and Host Country Characteristics in FDI: Firm- Level Evidence from Japan, Korea and Taiwan	2010
266	Abu S SHONCHOY	Determinants of Government Consumption Expenditure in Developing Countries: A Panel Data Analysis	2010
265	Dil Bahadur Rahut, Iván Velásquez Castellanos and Pravakar Sahoo	COMMERCIALIZATION OF AGRICULTURE IN THE HIMALAYAS	2010
264	Zhang Yu	Yangtze River Delta's System Integration: Institutional Barriers and Countermeasures	2010
263	Yuichi WATANABE	Tax Differentials and Inflow of Foreign Direct Investments: Evidence from Foreign Operations of U.S. Multinational Companies	2010
262	Yasushi HAZAMA	The Making of a State-Centered "Public Sphere" in Turkey: A Discourse Analysis	2010
261	Pravakar Sahoo, Ranjan Kumar Dash and Geethanjali Nataraj	Infrastructure Development and Economic Growth in China	2010
260	Takeshi KAWANAKA	The Urban Middle Class in the Instability of New Democracies	2010
259	Jose Ramon Albert, Soya Mori, Celia Reyes, Aubrey Tabuga, and Tatsufumi Yamagata	Income Disparity among Persons with Disabilities Assessed by Education and Sex: Findings from a Field Survey Conducted in Metro Manila, the Philippines	2010
258	Koji KUBO, Nu Nu Lwin	Smuggling and Import Duties in Myanmar	2010
257	Koichi FUJITA, Tamakai ENDO, Ikuko OKAMOTO, Yoshihiro NAKANISHI, Miwa YAMADA	Myanmar Migrant Laborers in Ranong, Thailand	2010
256	Dil Bahadur Rahut, Ivan Velasqez Castellanos, Pravakar Sahoo	Performance of Financial Institutions in Bhutan	2010
255	Mitsuhiro KAGAMI	Recent Trends in Asian Integration and Japanese Participation	2010
254	Hisaya ODA, Yuko TSUJITA	The Determinants of Rural Electrification in Bihar, India	2010
253	Kiyoyasu TANAKA, Naomi HATSUKANO	The Size Distribution of All Cambodian Establishments	2010
252	Kazunobu HAYAKAWA, Fukunari KIMURA, Tomohiro MACHIKITA	Globalization and Productivity: A Survey of Firm-level Analysis	2010
251	Kuo-I CHANG, Kazunobu HAYAKAWA, Toshiyuki MATSUURA	Location Choice of Multinational Enterprises in China: Comparison between Japan and Taiwan	2010
250	Kazunobu HAYAKAWA, Kiyoyasu TANAKA, Yasushi UEKI	Are Trading Partners Complementary in International Trade?	2010
249	Takeshi INOUE, Shigeyuki HAMORI	How Has Financial Deepening Affected Poverty Reduction in India? Empirical Analysis Using State-Level Panel Data	2010
248	Hisatoshi HOKEN	Restoration of Micro Data of John Lossing Buck's Survey and Analysis of the Inverse Relationship between Yield and Farm Size in Rural China in the 1930's	2010
247	Quoc Hung NGUYEN	Liability Dollarization and Fear of Floating	2010
246	Hideki HIRAIZUMI	Trade of Heilongjiang Province (China) with Russia	2010
245	Abu S SHONCHOY	The Dynamics of Spending and Absorption of Aid: Panel Data Analysis	2010
244	KHOO Boo Teik	Cyber-networks, physical coalitions and missing links: Imagining and realizing dissent in Malaysia 1998–2008	2010

No.	Author(s)	Title	
243	Miwa TSUDA	Kenya's 2007 Election Crisis	2010
242	Takeshi INOUE	Effectiveness of the Monetary Policy Framework in Present-day India Have Financial Variables Functioned as Useful Policy Indicators?	[:] 2010
241	Kiyoyasu TANAKA	Transport Costs, Distance, and Time: Evidence from the Japanese Census of Logistics	2010
240	Takeshi KAWANAKA	Polical Institutions and Policy Outcomes: Effects of Presidential Vetoes on Budget Making	2010
239	KHOO Boo Teik and Vedi R. HADIZ	Critical Connections: Islamic Politics and Political Economy in Indonesia and Malaysia	2010
238	KHOO Boo Teik	Social Movements and the Crisis of Neoliberalism in Malaysia and Thailand	2010
237	Ikuo KUROIWA Hiromichi OZEKI	Intra-regional Trade between China, Japan, and Korea: Before and After the Financial Crisis	2010
236	KHOO Boo Teik	No Insulation: Politics and Technocracy's Troubled Trajectory	2010
235	Koichi KAWAMURA	Is the Indonesian President Strong or Weak?	2010
234	Toshiyuki MATSUURA, Kiyoyasu TANAKA, Shujiro URATA	The Determinants of Offshore Production by Multinational Corporations (MNCs): A Comparison of Japanese and US MNCs	2010
233	Takeshi KAWANAKA	Interaction of Powers in the Philippine Presidential System	2010
232	Takahiro FUKUNISHI	FDI and Export particiaption of Local Firms in Africa: The Case of the Kenyan Garment Industry	2010
231	Hitoshi SUZUKI	A Critical Review of Opinion Polls relating to Iranian Voting Intentions: Problems of Research Methodology as applied to Complex Societies	2010
230	Mai FUJITA	The Diversity and Dynamics of Industrial Organisation: Transformation of Local Assemblers in the Vietnamese Motorcycle Industry	2010
229	Miki HAMADA, Masaru KONISHI	Related Lending and Bank Performance: Evidence from Indonesia	2010
228	Hisao YOSHINO	Strategic Trade Policy and Non-Linear Subsidy	2010
227	Masahiro KODAMA	Large Fluctuations in Consumption in Least Developed Countries	2010
226	Chiharu TAMAMURA	Cost Reduction Effects of "pseudo FTAs" in Asia - Application of a Price Model Based on a Multilateral I/O Table -	2010
225	Koji KUBO	Natural Gas Export Revenue, Fiscal Balance and Inflation in Myanmar	2010
224	Mariko WATANABE	Separation of Control and Lash-flow Rights of State Owned Listed Enterprises: Channels of Expropriation following Discriminated Share Reform	n 2010
223	Haruka I. MATSUMOTO	The Taiwan Strait Crisis of 1954-55 and U.SR.O.C. Relations	2010
222	Miwa TSUDA	The Experience of National Rainbow Coalition (NARC): Political Parties inKenya from 1991 to 2007	2010
221	Kensuke KUBO	Inferring the Effects of Vertical Integration from Entry Games: An Analysis of the Generic Pharmaceutical Industry	2010
220	Ikuo KUROIWA Hiroshi KUWAMORI	Shock Transmission Mechanism of the Economic Crisis in East Asia: An Application of International Input-Output Analysis	2010